

Deep seabed mining: Frontiers in engineering geology and environment

Xingsen Guo^{1,2} · Ning Fan^{3,4} · Yihan Liu¹ · Xiaolei Liu^{1,5} · Zekun Wang¹ · Xiaotian Xie¹ · Yonggang Jia^{1,5}

Received: 25 December 2022 / Revised: 28 January 2023 / Accepted: 1 March 2023 $\ensuremath{\textcircled{}}$ The Author(s) 2023

Abstract

Ocean mining activities have been ongoing for nearly 70 years, making great contributions to industrialization. Given the increasing demand for energy, along with the restructuring of the energy supply catalyzed by efforts to achieve a low-carbon economy, deep seabed mining will play an important role in addressing energy- and resource-related problems in the future. However, deep seabed mining remains in the exploratory stage, with many challenges presented by the high-pressure, low-temperature, and complex geologic and hydrodynamic environments in deep-sea mining areas, which are inaccessible to human activities. Thus, considerable efforts are required to ensure sustainable, economic, reliable, and safe deep seabed mining. This study reviews the latest advances in marine engineering geology and the environment related to deep-sea mining activities, presents a bibliometric analysis of the development of ocean mineral resources since the 1950s, summarizes the development, theory, and issues related to techniques for the three stages of ocean mining (i.e., exploration, extraction, and closure), and discusses the engineering geology environment, geological disasters, in-situ monitoring techniques, environmental protection requirements, and environmental effects in detail. Finally, this paper gives some key conclusions and future perspectives to provide insights for subsequent studies and commercial mining operations.

Keywords Deep seabed mining \cdot Marine engineering geology \cdot Geological disasters \cdot Environment \cdot Techniques

1 Introduction

Ocean mining has been developed since the 1950s, and deep seabed mining has seen a boom in the last decade. Many high-value deep-sea minerals have been discovered in a vast deep-sea area of 360 million km² with a water depth of more than 200 m (Miller et al. 2018). As the most valuable mineral

⊠ Xiaolei Liu xiaolei@ouc.edu.cn

Yonggang Jia yonggang@ouc.edu.cn

> Xingsen Guo xingsen.guo@ucl.ac.uk

Ning Fan fanning@wzu.edu.cn

Yihan Liu liuyihan0926@stu.ouc.edu.cn

Zekun Wang zkwang0324@stu.ouc.edu.cn

Xiaotian Xie 17615219599@163.com resources of this century, deep-sea minerals are critical for high-tech industries related to the low-carbon economy (e.g., new forms of transportation, aerospace components, electric vehicles, renewable energy, and electrical storage).

Deep-sea minerals mainly include gas hydrates, which is a crystalline solid formed of water and gas, and polymetallic nodules, cobalt-rich crusts, polymetallic sulfides,

- ¹ Shandong Provincial Key Laboratory of Marine Environment and Geological Engineering, Ocean University of China, Qingdao 266100, China
- ² Department of Civil, Environmental, and Geomatic Engineering, University College London, London WC1E 6BT, UK
- ³ College of Civil Engineering and Architecture, Wenzhou University, Wenzhou 325035, China
- ⁴ Key Laboratory of Engineering and Technology for Soft Soil Foundation and Tideland Reclamation of Zhejiang Province, Wenzhou University, Wenzhou 325035, China
- ⁵ Laboratory for Marine Geology, Qingdao National Laboratory for Marine Science and Technology, Qingdao 266061, China

and rare earth elements, which are rich in Mo, Mn, Se, Ni, Au, and Cu, have the potential to replace terrestrial metallic resources and have high commercial exploitation value (Pak et al. 2018; Liu et al. 2021a). Figure 1 presents the global sea distributions of polymetallic nodules, cobalt-rich crusts, and polymetallic sulfides along with their occurrence states and representative images. Polymetallic nodules are distributed in deep-sea plains at water depths of 4000-6500 m or fully or partially buried in sediment. The main industrial exploration areas of polymetallic nodules include the Eastern Pacific, the Peruvian Basin, and the Northern Indian Ocean Center (Pan and Hua 1996; Sharma 2017). Cobalt-rich crusts are distributed on the tops, flanks, and broad saddle structures of seamounts with water depths of 800–3000 m and connected to the hard bedrock surface. These crusts are mainly concentrated in the West and Central Pacific (Sharma 2017; Wei et al. 2017). Polymetallic sulfides are widely distributed in the hydrothermal zones of mid-ocean ridges, volcanic arcs, and back-arc spreading centers with water depths of 1000-3500 m. These sulfides occur in the Red Sea and Western Pacific (Ding et al. 2009; Liu et al. 2022). Given their areas of distribution, it is not easy to exploit these mineral resources.

Driven by the combination of industrialization and new energy development, companies from many countries have been scrambling to conduct mining trials on the ocean floor to obtain high-value minerals. In the 1950s, the United States, West Germany, United Kingdom, France, Union of Soviet Socialist Republics, and Japan began to investigate polymetallic nodules. In the 1970s, the multinational companies Ocean Mining Inc., Ocean Mining Associates, and Ocean Minerals Company, which are composed of several companies from the United States, Germany, the Netherlands, Belgium, and Italy, carried out polymetallic mining tests from 1978 to 1979, resulting in great progress (Chung 1998; Xiao et al. 2000). In 1994, the International Seabed Authority (ISA) was established to organize all mineral-related activities in the international seabed area for the joint management and development of resources. ISA approved "Regulations on prospecting and exploration for cobalt-rich ferromanganese crusts in the Area" in 2009, "Draft regulations on



Fig. 1 Global distribution, occurrence state, and representative images of three major deep seabed mineral resources (modified from Miller et al. 2018; Cuyvers et al. 2018; photos taken by ROV KIEL

6000 during cruise M78/2, Jan Steffen/GEOMAR, and GEOMAR/ Schmidt Ocean Institute/CSSF)

prospecting and exploration for polymetallic sulphides in the Area" in 2010, and "Regulations on Prospecting and Exploration for Polymetallic Nodules in the Area" in 2013 (International Seabed Authority 2009, 2010, 2013; He et al. 2016). In 2020, the American Bureau of Shipping issued the "Guide for Subsea Mining," the first seabed mining guidelines for the marine industry. This guide provides requirements for the design, construction, installation, and survey of mobile offshore mining units. Along with increasing environmental awareness, the impact of mining activities on the environment has attracted increasing attention. The Legal and Technical Commission of ISA issued "Draft Regulations on Exploitation of Mineral Resources in the Area" in 2017, with subsequent revisions released in 2018 and 2019 (Wang et al. 2022). In 2021, the China Association of Oceanic Engineering approved the "Guidelines for the Environmental Protection and Preservation of Deep-sea Mining Activities (China Association of Oceanic Engineering 2021)," which describes general principles for deep-sea mining activities. Table 1 lists mining trials of deep seabed minerals and operational tests for mining equipment from 1978 to 2019.

While deep seabed mining has great potential, it remains in the exploratory stage, with the associated technology still being developed. Unlike mining operations on land (Sasaoka et al. 2016; Lian et al. 2020; Zhang et al. 2022), deep seabed mining is complicated by high-pressure and low-temperature environments with complex engineering geology and hydrodynamics. Deep-sea areas are not reached by human activities, thereby requiring the simultaneous development of science (technologies) and human cognition. Because deep seabed mining systems involve complex scientific backgrounds (e.g., multi-disciplinary) and challenging engineering problems, this study discusses the engineering geology and environmental issues related to deep seabed mining to support subsequent studies and commercial mining operations.

The remainder of this paper is organized as follows. Section 2 presents a systematic bibliometric analysis of the development of ocean mineral resources. Section 3 discusses the development, theory, and issues related to the three stages of ocean mining activities (i.e., exploration, extraction, and closure) in detail. Section 4 provides an indepth discussion of the engineering geology environments and geological disasters at deep seabed mining sites along

Time (year)	Country/Company	Water depth (m)	Project content
1978	Ocean Management Inc (OMI)	5500	А
1978	Ocean Mining Association (OMA)	4570	А
1979	Ocean Mining Corporation (OMCO)	5000	А
1989	Germany/the former Ministry of Science and Technology of the Federal Republic of Germany	4140-4180	В
1990	Russia/Moscow Institute of Geological Exploration	79	С
1996	India/National Institute of Ocean Technology (NIOT); Germany/University of Siegen	500	D and E
1997	Japan/Polymetallic nodules Mining system research and development project	2200	F
1998	Japan/the Geological Survey of Japan, National Institute of Advanced Industrial Science and Technology (GSJ-AIST)	1600	G
2002	Japan/Japan Oil, Gas and Metals National Corporation (JOGMEC)	1600	D
2006	India/NIOT	450	F
2009	Korea/Korea Institute of Geoscience and Mineral Resources (KIGAM)	100	Н
2012	Japan/JOGMEC	1600	Е
2013	Korea/Korea Institute of Ocean Science and Technology (KIOST)	1370	F
2015	Korea/KIOST; Korea/Korea Research Institute of Ships and Ocean Engineering (KRISO)	1200	С
2016	China/Changsha Research Institute of Mining and Metallurgy	300	Н
2017	Japan/JOGMEC	1600	C and E
2018	China/Changsha Research Institute of Mining and Metallurgy	500	Е
2018	Netherland/Royal IHC	300	D
2019	China/Institute of Deep-sea Science and Engineering, Chinese Academy of Sciences (IDSSE,CAS)	2500	F

Table 1Mining trials of deep seabed minerals and tests of some mining systems (modified from Thiel and Schriever 1990; Ding et al. 2009;
Yang et al. 2020; Su 2020)

Notes: A, Mining trial for polymetallicnodules; B, Disturbance and recolonization experiment for deep seabed sediments; C, Test of the hydraulic lifting system; D, Test of the travel performance of the mining vehicle; E, Test of the acquisition performance of the mining vehicle; F, Test of the mining vehicle; G, Economic and technical feasibility study of polymetallic sulfide mining; and H, Test of the delivery system of the mining vehicle with the corresponding in-situ monitoring techniques. Section 5 discusses the requirements for environmental protection along with the environmental effects of deep seabed mining with a focus on the sediment plume. The summary and conclusions are presented in Sect. 6

2 Bibliometric analysis of ocean mineral resource development

Through bibliometric analysis, the development trends of some key information (e.g., literature numbers, authors, and keywords) from research literature can be assessed in a more objective and reliable way using mathematical and statistical theories. Here, the bibliometric analysis tool, CiteSpace-6.1.R2 (Chen 2004, 2006), was employed to summarize the overall status of the research relevant to ocean mineral resources and investigate the research development trends. The research literature data were retrieved from the Web of Science (WOS) database and the China National Knowledge Infrastructure (CNKI) database. Original data, including titles, authors, keywords, abstracts, etc., were output from these databases. The subject words used in the retrieval were "Ocean (or marine, undersea, underwater) + Mineral resources (or mining)". As a result, a total of 5838 effective literature records were obtained, including 4986 literature from the WOS database and 852 literature from the CNKI database, in a range of publication years from 1957 to 2022. The time of data retrieval was July 2022. The information from that 5838 literature was imported into the tool CiteSpace-6.1.R2 as basic data for bibliometric analysis.

2.1 Trends in the number of literature

The number of literature published in a certain time can reveal development trends in a research field to a certain extent. Figure 2 shows the variation trends of 5838 relevant literature in the field of ocean mineral resources over 65 years from 1957 to 2022 (where two databases are displayed separately). The time period of the analysis was divided into 12 time spans, with one span consisting of five years (except for the first and last spans).

Overall, the number of published literature related to ocean mineral resources gradually increased over time from 1957 to 2022. According to the variation trends, the development in the field of ocean mineral resources can be roughly divided into three stages: Stage I (1957 to 1988), Stage II (1989 to 2008), and Stage III (2009 to 2022). In Stage I, the average number of literature in each time span was approximately 19 for the WOS database and 22 for the CNKI database. These low literature numbers indicate that ocean mineral resources did not attract wide interest from experts and engineers during that time. Besides, the





Fig. 2 Trends in the number of relevant literature published from 1957 to 2022

low literature numbers are also related to the economic and social development status at that time. In Stage II, the research situation had greatly improved, with ~ 280 studies in each time span for the WOS database and ~ 40 for the CNKI database. Compared to Stage I, the number of WOS database literature was nearly 15 times higher in Stage II. In Stage III, the number of relevant studies again increased substantially, with the average number of literature in each time span reaching ~ 1256 for the WOS database and ~ 195 for the CNKI database. Those trends highlight how ocean mineral resource development has become a hot research topic in the more than past 10 years. In addition, the trends could herald that this research topic will remain popular for a long time in the future.

2.2 Degrees of national concern and international cooperation

The institutional information extracted from research literature can inform the concern and cooperation between countries or regions in a research field to a certain extent. According to the data extracted from 4986 literature in the WOS database, a network graph of national concern and cooperation in research on ocean mineral resources was constructed (see Fig. 3). The clusters in the figure represent the countries of the institutions conducting the research, and the links between the clusters represent cooperation between institutes in different countries. Statistical analysis revealed 135 clusters and 163 links (not all clusters and links are displayed in Fig. 3 due to interface limitations).



Fig. 3 Network plot of national concern and international cooperation based on the WOS database

The results in Fig. 3 indicate cooperation from 135 countries or regions on the topic of ocean mineral resources, and the degree of cooperation was high (the ratio of links to clusters was 163/135 = 1.21). The three countries with the most relevant literature are the United States, China, and Germany, which show large clusters. Those highly concerned countries or regions often have longer shorelines and need sufficient research to support their ocean mineral resources development, as shown in Fig. 4 (here, the data from all countries were grouped into continents, and the number of published studies is plotted with shoreline length by continent; The shoreline length data are based on Liu et al. (2015)).

2.3 Trends in literature keywords and hotspots

Literature keywords can intuitively reflect the theme of a study and affect the findings of literature searchers and whether the reader is interested in reading the study. Figure 5 shows clusters representing the literature keywords for studies on ocean mineral resources from the WOS and CNKI databases. The clusters represent the occurrence frequency of one keyword in all literature data, and the links between the clusters represent the correlations between keywords. As shown in Fig. 5a, the five most common literature keywords from the WOS database were "heavy metal", "sediment", "trace element", "contamination", and "rare earth element"



Fig. 4 Number of relevant literature versus shorelines by continent



Fig. 5 Literature keywords on ocean mineral resources from the a WOS database and b CNKI database

(sorted from more frequent to less frequent); the five most common literature keywords from the CNKI database were "polymetallic nodule", "marine sediment", "flooded mine", "mining vehicle", and "environmental impact" (Fig. 5b). The popular research directions indicated by the keywords from the two databases are similar and include the mineral resources themselves (e.g., metals and nodules), resource development approaches, and the environmental effects of resource development. Those research directions clearly align with the full research process from identifying resources to safely developing the resources. In addition, some keywords were often mentioned together, forming the keyword links in Fig. 5. For example, "heavy metal" and "contamination" are linked in Fig. 5a, while "polymetallic nodule" and "mining vehicle" are linked in Fig. 5b.

To further investigate the frequencies of literature keywords used over time, the top 10 hotspots in the literature on ocean mineral resources over time are shown in Fig. 6. Through bibliometric analysis, the hotspots were counted in a way that a keyword in a certain period suddenly increases by a big margin. The larger the degree of increase, the higher the "strength" value in Fig. 6. In addition, the total time period of 1957 to 2022 was also divided into three stages (Stage I, 1957 to 1988; Stage II, 1989 to 2008; and Stage III, 2009 to 2022) as introduced in Sect. 2.1. As shown in Fig. 6a, for the WOS database, no obvious hotspots were found in Stage I. In comparison, Stage II had more hotspots, and they remained for a long time. Among the keywords in this stage, "sediment" had the largest strength value of 24.85. This indicates that the formation, distribution, and development of ocean mineral resources are inseparable from the study of seabed sediments. While in Stage III, the hotspots in this stage were related to the process of deep-sea resource development (e.g., extraction techniques and extraction effects on biosynthesis). The CNKI database also shows a similar development trend as the WOS database, as shown in Fig. 6b.

3 Techniques for ocean mining

Marine sediment

Polymetallic nodule

Ocean science

Critical metal

1957-2022

1957-2022

1957-2022

1957-2022

2.11

2.27

2.34

1.56

2013

2016

2018

2018

2016

2022

2019

2020

In general, ocean mining activities have three main stages according to their order of implementation: exploration, extraction, and closure (China Association of Oceanic Engineering 2021). The exploration stage refers to the period of an ocean mining project after an exploration permit has been granted but no mining permit. The geologic and hydrologic environmental conditions, geotechnical properties of the seafloor soils, potential hazards, and relevant suggestions for the project should be provided during the exploration stage. After the mining permit is granted, the project enters the extraction stage, and mining extraction activities can be organized and carried out. When the mining task is completed or stopped, the project enters the closure stage, in which the ecosystem of the mining area is restored. The subsequent sections introduce the techniques involved in each phase of ocean mining.

3.1 Techniques for the exploration stage

Comprehensive exploration of an ocean mining project should cover the ocean life, meteorology, chemistry, and geology based on techniques from fields including engineering geology, geotechnical engineering, and biology. This study focuses on techniques related to geological exploration, which primarily include geophysical exploration and geotechnical exploration.

3.1.1 Geophysical exploration techniques

Geophysical exploration can reveal the geological characteristics (e.g., geological background, stratigraphic substance



Fig. 6 Top 10 hotspots in the literature on ocean mineral resources during different time periods a WOS database and b CNKI database

(b)

composition, and geological structures) of the project area based on the local variation in geophysical fields. Geophysical exploration mainly considers the features of subsea stratigraphic substances (e.g., the density, elasticity, magnetic permeability, electrical conductivity, thermal conductivity, and radioactivity) to elucidate and explain the variation in geophysical fields. Thus, geophysical exploration is an indirect method. According to the features being detected, common geophysical exploration techniques include gravity exploration, seismic exploration, magnetic exploration, and electrical exploration.

(1) Gravity exploration

The principle of gravity exploration is based on the local variation in the gravitational acceleration fields of the seafloor caused by the density differences of various rocks and ores, as shown in Fig. 7. The local variation of the gravitational acceleration field (i.e., gravity anomaly) can be explored by gravity devices, mainly shipborne equipment (e.g., gravimeters and torsion balances). The plans and sections of the gravitational acceleration field can then be obtained and used to infer the position of mineral resources. Gravity exploration can be performed using fixed-point surveys or regional continuous surveys.

The gravity exploration technique was developed in the early twentieth century. Baron Roland von Eotvos made the first gravity field measurement in 1901 in a subaquatic environment (Balaton lake) using a self-developed torsion balance for measuring gravity variation (Vajk 1949).



Fig. 7 Example of the gravity exploration technique (modified from Essa et al. 2020)

Subsequently, the torsion balance was employed at sea by Hecker in 1903 (Torge 1989). A gravity device was first attached to a submarine for exploration by Vening Meinesz in 1923 (Fujimoto et al. 2000). revealing changes in gravity in different oceanic regions; in particular, a large gravitational effect was found near the Indonesian Java Trench. Later, in the 1950s, gravity exploration approaches were widely applied in many countries, and the exploration efficiency was continuously improved. Some instruments used for gravity exploration include the pendulum ocean gravimeter of Graf-Askania and LaCoste & Romberg and the vibration ocean gravimeter of the University of Tokyo and the Massachusetts Institute of Technology (Forsberg et al. 2015; Shinohara et al. 2015). Since the development of global positioning system (GPS)-based 3D positioning, the gravity exploration approaches have been greatly improved. The gravity anomalies caused by resource storage can be detected more accurately with the support of GPS. Recently, some undersea remotely operated vehicles (ROVs) or autonomous underwater vehicles with gravity exploration functions have been developed (Ballu et al. 1998; Gilbert et al. 2007; James et al. 2013).

(2) Seismic exploration

The principle of seismic exploration is based on the propagation of seismic waves in strata with different elasticities. First, the seismic waves are artificially excited (e.g., by an explosive source, air gun, spark detonation gas, or resonator). The reflected or refracted waves at the interfaces between different elastic strata are received and recorded by hydrophones. Finally, the strata are identified by the different wave travel times and spectra. Seismic exploration is often used to explore oil and gas reservoir layers, coal fields, salt rock deposits, and some metallic mineral deposits. This technique can be used for continuous observation at sea and reach exploration depths of over a thousand kilometers, resulting in high efficiency.

In 1845, Mallet used artificially excited seismic waves to measure the propagation speed of elastic waves in the Earth's crust, which is considered the birth of seismic exploration (Graziano and Anita 2005). Regarding the application of seismic exploration, a team of physicists and geologists (William P. Haseman, J. Clarence Karcher, Irving Perrine, and Daniel W. Ohern) first recorded clear reflection waves produced by an artificial seismic source in Oklahoma (United States) in 1921 (Schriever 1952). Three oil storage fields were then discovered in the area through reflected seismic exploration in 1930 (Schriever 1952). During the same period, many salt domes were discovered along the Gulf Coast using refracted seismic surveys (Sussman and Stallman 1975). In the 1950s and 1960s, the spot photography technique of seismic exploration was replaced by analog tape recording, thereby improving the signal recording quality (Luo 2009). Later in the 1970s, analog tape recording was replaced by digital tape recording, resulting in a complete system based on highspeed digital computers that combined digital recording, multiple overlay techniques, and seismic data processing. In the 1990s, a new seismic exploration technique, seismic imaging, was proposed. The method utilized the characteristics of no surface wave interference in water and forms color digital soil sections by means of densely displaying wave impedance (Mora and Tarantola 1989). Recently, broadband seismic exploration techniques have been proposed for exploration in areas with large variations in water depth, rugged seabeds, and developed fractures (Moldoveanu et al. 2012).

From the perspective of testing dimension, seismic exploration approaches have developed from 1D to 4D (i.e., from the exploration of a point, to a line, to an area, and to multiple areas at different times). Time-varying 3D

stereoscopic seismic images of the seabed layers can now be obtained using 4D seismic exploration approaches. In addition, hydrophone units have been placed on the seabed for long-term exploration, a technique known as ocean bottom seismometers (OBS) (Jocobson et al. 1991), as shown in Fig. 8. The OBS exploration technique allows the seabed reception of seismic waves, thus avoiding the large amount of absorption and attenuation of seismic wave energy in seawater and improving the accuracy of the results (Zhang 2015).

(3) Magnetic exploration

The principle of magnetic exploration is based on the local anomalies in the geomagnetic fields caused by various magnetic rocks, ores, and other objects (Fig. 9a). A magnetic field detector (e.g., proton precession magnetometer and gradiometer) is first fixed on a subsea drag body. The body is then dragged within a certain distance behind

Fig. 8 a OBS principle and **b** example output (modified from Sha et al. 2015)







(b)

Fig. 9 Magnetic exploration a Working principle (modified from Hsu et al. 2014), where W is the weight as a depressor unit (The towed electromagnetic transmitter composed of 3 receivers (Rx1, Rx2, and Rx3, tuned to measure these magnetic fields different frequencies) and a transmitter (a horizontal magnetic dipole) forming an array with a total length of 40 m on the seafloor generates harmonic magnetic fields over a range of frequencies (Hsu et al. 2014)) b Example output on land (modified from Kim 2021) and c Example output on the exploration of underwater natural gas resources (modified from Evans 2007)



a moving ship, and the local magnetic anomalies of the stratum are recorded. This technique is suitable for exploring ore resources (e.g., iron ore, lead–zinc ore, and copper ore), Fig. 9b presents a example output on land, and Fig. 9c

presents a recorded example on the exploration of underwater natural gas resources. However, this technique involves intermittent measurements, and the survey sensitivity is relatively low.

Magnetic exploration is the earliest developed and most widely used among all geophysical exploration approaches, and some magnetite can be easily found using a magnetic compass. In 1870, Thaln and Tiberg constructed a universal magnetometer, and the magnetic exploration approach was formally proposed (Zhang 2011). Schmidt invented a quartz blade magnetometer to find minerals other than iron ores in 1915. In the 1950s and early 1960s, magnetometers were installed on ships to conduct magnetic surveys of oceans (Howland-Rose et al. 1980). Later, a transient electromagnetic approach was developed with a deeper exploration depth and strong penetration in high-resistance layers. In the 1980s, the rapid development of various computer technologies and high-precision magnetometers promoted the development of magnetic exploration based on high-precision and automatic interpretation (Chen 2015). The inversion visualization technique for interpretation has been developed and realized (Zhang 2011).

(4) Electrical exploration

The principle of electrical exploration is based on the differences in the electrical properties (e.g., electrical conductivity, electromagnetic induction, and dielectricity) of rocks and ores (Fig. 10). Here, fields are of purely radial geometry and able to generate galvanic effects when they intersect subhorizontal, tabular bodies (Constable 2010). The implementation of this technique is similar to the implementation of magnetic exploration, except that the field sources in electrical exploration are electrical signals with different frequencies and waveforms. The spatial distributions of the electric and electromagnetic fields on the seafloor are then recorded and analyzed. Based on the current properties of the field source, the electrical exploration technique can be further divided into the direct current method and the alternating current method.

In the early twentieth century, Schlumberger conducted offshore measurements of direct current resistivity, marking the beginning of electrical exploration in the ocean (Allaud and Martin 1977). Corwin (1976) discovered a 300-mv self-potential anomaly in Penobscot Bay, Myanmar using the self-potential method (a type of direct current method) and found a sulfide sedimentary deposit. Goto et al. (2008) developed MANTA, an ocean direct current resistivity measurement system for probing the top boundaries of methane hydrate fields; MANTA was successfully used to detect methane hydrate resources near Okinawa in Japan. Recently, using statically placed electrodes and drag electrode techniques, Simyrdanis et al. (2016) performed 2D and 3D direct current inversion to image the seafloor layers near the coast of northwestern Greece.



Fig. 10 Electrical exploration **a** Working principle (i.e. the dipole geometry of a near-seafloor transmitter, here the maximum vertical electric fields (red) are below the transmitter in the inline direction, and fields are of purely radial geometry and able to generate galvanic effects when they intersect subhorizontal, tabular bodies such as oil and gas reservoirs (Constable 2010)) and **b** Example output (Constable 2010), where Tx and Rx represent transmit and receive, respectively

3.1.2 Geotechnical exploration techniques

Compared to the large-scope survey of geophysical exploration, geotechnical exploration is based on a fixed-point survey. Geotechnical exploration mainly serves to test and evaluate the geological conditions (e.g., composition, structure, formation age, and genetic type of each soil layer along with adverse geologic actions) and the geotechnical properties of the seafloor soils (e.g., particle size distribution, void ratio, liquid and plastic limits, compression modulus, cohesion, and internal friction angle) in the project area. This information can be used to judge the stability and suitability of the project site. Geotechnical exploration is a direct method. According to the test site, the techniques for geotechnical exploration include in-situ and laboratory testing.

(1) In-situ testing

In-situ testing is carried out directly on the seabed of the project area, where the data are closer to the real state. In-situ testing techniques for geochemical exploration are relatively simple and depend on the working ocean environment. The in-situ testing process typically involves the vertical penetration of various probes (e.g., cone type, cross-plate type, dish type, pillar type, T-bar type, and ball type) into the seafloor soil layers. The soil layer information is then inferred according to the penetration resistance or rotational torque force. Implementing the test relies on diving devices (e.g., cabled work platforms and ROVs) and power devices (e.g., chain-driven penetration systems and gear-driven penetration systems). Figure 11 shows some common in-situ testing techniques.

There are many types of in-situ testing techniques. According to current engineering survey reports of ocean wind farm projects and oil and gas pipeline projects, commonly used testing techniques include cone penetration test (CPT), standard penetration test, dynamic penetration test (DPT), vane shear test (VST), and full-flow penetrometer test (e.g., using T-bar and ball probes). VST, the first technique to be used in ocean areas, was first applied over 100 years ago (Cadling and Odenstad 1950; Briaud 2013). The principle of VST is relatively simple, and few influencing factors are involved (Zhang et al. 2021a). While VST allows the shear strengths of seabed soils to be obtained quickly, it cannot achieve continuous testing along the depth direction of the soil layers. In the 1930s, the concept of CPT was developed, with the most representative CPT device being the Dutch cone (Stanton et al. 1939; Seed and De Alba 1986). Begemann further developed CPT in the 1950s (Begemann 1953) to obtain the frictional resistance of the penetration bar. CPT then began to be used for the engineering classification of soils. CPT was first used for underwater testing in the 1960s and extended to deep-sea testing in the 1980s (Hanzawa and Tanaka 1992). In 2010, MARUM developed GOST, a commercial CPT system that functions in water depths up to 4000 m. Current CPT probes can carry numerous functional sensors, including pore pressure sensors, resistivity sensors, and wave transceivers (Lunne et al. 2009; Zou et al.2020; Liu et al. 2021b).

Unlike CPT, DPT requires additional penetration power; thus, free-fall DPT penetrometers are mainly used for ocean soil testing. The first ocean free-fall DPT penetrometer was



Fig. 11 Common in-situ testing techniques (modified from Mayne et al. 2009)

developed by the Memorial University of Newfoundland in 1973; this device had a limited testing water depth of 180 m and soil penetration depth of 4.5 m (Dayal and Allen 1975). In 2006, MARUM also developed a DPT system (DWFF-CPTU, Deep Water Free Fall Piezocone penetration Testing) (Lykousis et al. 2007). The maximum working water depth reached 4000 m, allowing the system to support deep-sea exploration. Full-flow penetrometer testing represents an improvement on CPT. The probes adopt T-bar and ball shapes, providing more accurate soil strength results. The full-flow penetrometer technique was first proposed and developed in the early 1990s by scholars at the Centre for Offshore Foundation Systems (Stewart and Randolph 1991; Fan et al. 2020a). Subsequently, Low et al. (2008) proposed a new manually operated full-flow penetrometer system that is easy to use on board. Recently, Guo et al. (2022a) developed a comprehensive seabed soil testing system equipped with full-flow penetrometers (i.e., ball and streamline probes), CPT, and VST probes. This system was used to evaluate the mechanical properties of sediments in the South China Sea (Guo et al. 2021a, b, 2023a; Gu et al. 2022).

(2) Laboratory testing techniques

To eliminate interfering factors from the project site, representative soil samples from the project site can be tested in the laboratory (sometimes directly in the ship laboratory) using more precise and sensitive instruments than can be applied in the field. It should be noted that the quality of the soil samples directly affects the value of the test results. Subsea sampling is generally conducted by gravity sampling or drilling sampling. In gravity sampling, the samplers (e.g., surface grab samplers, which are generally used for sampling depths < 1 m, and open-column samplers, which are generally used for sampling depths < 6 m) freely drop to the seabed under the influence of gravity and are dragged back using a cable. In drilling sampling, holes are first drilled in the seabed, and a thin-walled sampler is then used to extract soil samples from the drill hole. The sampling depth for drilling sampling can reach ~ 60 m. Laboratory testing can then be used to obtain the geotechnical properties (i.e., physical, chemical, mechanical, and microcosmic indexes) of the seabed soils based on soil mechanics theories. Figure 12 shows some common laboratory testing techniques.



Fig. 12 Common laboratory testing techniques (modified from Mayne et al. 2009)

Most laboratory testing techniques for marine soils follow traditional soil tests, which are not introduced here. It is worth mentioning that the construction and development of submarine laboratories have accelerated in recent years. Submarine laboratories are designed according to the principle of saturation diving and can be moved or fixed. The working water depth of submarine laboratories has reached more than 300 m (Su 2020). If submarine laboratories become commonplace, geotechnical exploration with laboratory testing techniques will become even easier.

Regulations covering ocean engineering usually require geophysical exploration and geotechnical exploration to be conducted simultaneously so that comprehensive seafloor geological and geotechnical information can be obtained. Thus, there are many examples of applications that combine these two exploration techniques. For example, holes can be drilled first to obtain soil samples for laboratory testing followed by geophysical exploration (e.g., electrical and magnetic tests) in the drill hole.

3.2 Techniques for the extraction stage

The extraction stage is the primary component of ocean mining and requires various devices and systems to work together, including the mining ship or platform, launch and recovery systems, resource collection systems, dewatering and subsea mining cargo handling systems, and subsea mining control and monitoring systems (American Bureau of Shipping 2020). The techniques applied in the extraction phase often differ based on the physical form and depth of the resources being extracted. Overall, extraction techniques for ocean mineral resources can be divided into two types according to the systems used: mobile extraction techniques and fixed extraction techniques.

(1) Mobile extraction techniques

Mobile collection techniques are generally used for seabed surface resource extraction (resources with granular or nodular shapes, like polymetallic nodules). These techniques rely on ship-dragged collection devices (e.g., chain buckets, crab buckets, and pipe-suction ends) or self-propelled collection devices (e.g., subsea mining vehicles and miniature submarines) for ocean resource mining. The collection devices are recovered after the temporary storage devices are filled. Some common mobile collection techniques are summarized in Fig. 13.

The development of mobile extraction techniques is related to the water depth of the mining operation. For shallow sea resource mining (e.g., placer), most extraction techniques are derived from land mining techniques and use mining vessels such as chain bucket vessels, crab bucket vessels, and pipe-suction end vessels. The technical



Fig. 13 Example mobile extraction techniques (modified from Bath 1989)

principle of these techniques is relatively simple. Mineral resources are collected under the action of these dragged collection or suction devices. The cost of these extraction techniques increases as the water depth increases. The largescale extraction of deep seabed resources (e.g., polymetallic nodules) began in the 1970s (Bath 1989). In this system, the bucket is freely dropped from the vessel, and the worker on the vessel receives a signal when the bucket reaches the seabed. The bucket is then dragged using cables for collection. The continuous line bucket mining system was improved for use in deep-sea mining (Masuda et al. 1971). However, the system cannot adapt to the complex seabed topography, and the cables easily wind, resulting in low mining efficiency. In the 1980s, a shuttle vessel deep-sea mining system was developed by the AFERNOD in France (Gauthier and Charles 1979; Lenoble 1981; Du et al. 2016). This system requires a submarine to dive into the seabed for collection and can work at seabed depths up to 6000 m, with the daily collection amount reaching 2000 t. However, the continuous power issues in this system were difficult to solve. During the same time period, a pipeline lift mining system developed in the United States began to be used in deepsea environments (Chung 2003), resulting in large-scale mining and high mining efficiency. Subsequently, Grebe (1997) proposed a cooperative mining system based on a flexible transfer pipe and ore collector to adapt to the complex seabed terrain. More recently, subsea mining vehicles have emerged as a valuable tool for deep seabed resource extraction (Hong et al. 2019). These vehicles provide a flexible system with high mining efficiency and do not require a lifting system.

(2) Fixed extraction techniques

Fixed extraction techniques are required for some resources in deep seafloor layers or bedrock (e.g., oil, natural gas, combustible ice, brimstone, and coal). Fixed collection techniques first drill through the seabed to the resource storage area and reinforce the drill-hole wall to form stable extraction wellheads. Working tunnels may be further dug in some projects. The extraction devices (e.g., centrifugal pumps, boring bits, and jet devices) then provide flow pressure, shaking force, or high-temperature melting to release the resources, which are then transmitted to the platform or ship through conveyor belts, pipelines, or risers (Nian et al. 2018; Fan et al. 2022).

Fixed extraction techniques can be categorized based on their primary system and platform (see Fig. 14). The world's first offshore extraction well was drilled off the coast of California in the United States in 1897 (Silcox et al. 1987; Pan 2006) for the extraction of offshore oil. A wooden drilling platform with a large size and shallow operating level was first employed. Steel offshore platforms began to be used in the 1950s and 1960s, resulting in the rapid development of offshore oil and gas resources (Nian et al. 2022). By the late 1960s, the operating water depth had exceeded 200 m. In 1973, the first concrete platform was built in the Ekofisk oilfield in the North Sea (Aitcin and Houle 1986). In the 1980s, with the further development of deep-water oil and gas extraction, tension leg and guyed tower platforms were developed with working deeps of 600 m. At present, semisubmersible platforms are mainly used for mining in deepwater areas and can be used to extract multiple resources (e.g., oil, natural gas, combustible ice, brimstone, and coal). Representative systems include the Aker H-6 system from AKER Solutions in Norway, the DSS50 system from GustoMSC in the Netherlands, and the GVA7500M system from GVA in Sweden (Ted 2014).

3.3 Techniques for the closure stage

To decrease the impacts on ocean ecological environments as much as possible, environmental restoration remains critical during the closure stage after the mining activities are finished. The regulations on ocean mining in many countries put forward clear requirements for closure activities (American Bureau of Shipping 2020; China Association of Oceanic Engineering 2021). In addition to controlling the mining scale, the environmental impacts of mining, and the carbon footprint of extraction, the mine tailings (ore waste resulting from mining, typically a mud-like material) must



Fig. 14 Examples of resource extraction platforms (modified from Yu et al. 2015 and https://worldoceanreview.com/de/wor-3/oel-gas/wie-und-wo-gefoerdert-wird/)

be controlled and dealt with (e.g., by returning mine tailings back to the depth required by ISA or coastal states using the subsea mining downcomer and pumping system and returning tailings pipes for transporting seafloor mines from the seafloor to a mobile offshore mining unit).

After extraction, common ocean ecosystem restoration approaches include the construction of artificial algae farms, the planting of mangroves, and the placement of artificial reefs. Large-scale seaweed farming is one of the best ways to improve the marine ecological environment. Many algae (e.g., sargassum and stone lily) accumulate heavy metals and thus can purify seawater polluted with heavy metals (Phaneuf et al. 1999). Similarly, mangrove plants can absorb heavy metals in seawater and sediments through their roots, reducing the content of heavy metals in seawater (Ke et al. 2005). Artificial reefs can create a good environment for fish to inhabit and help restore sea areas with ecological imbalances and severe resource deficits (Fisher et al. 2015).

4 Engineering geology issues in deep seabed mining

4.1 Evaluation of the engineering geology environment

Evaluating the engineering geology environments of deep seabed mining sites is of great significance for preventing and controlling geologic hazards, protecting the environment, selecting the mining machinery and equipment, and designing the mining system (Sharma 2017). However, there are no mature procedures for the evaluation of the engineering geology environments of deep seabed mining sites. This section provides a comprehensive discussion of the engineering geology and environmental characteristics of mineral sites and briefly summarizes the process of evaluating the engineering geology environments of deep seabed mining sites.

In general, evaluating the engineering geology environment requires an on-site investigation, collection of corresponding data (including historical data), and development of an evaluation report (Guzzetti et al. 2005; Fell et al. 2008; Shano et al. 2020; Xie et al. 2022; Guo et al. 2023b, c; Shan et al. 2022; Dong et al. 2022, 2023). The evaluation report should describe the characteristics of the regional engineering geology environment as well as identify and evaluate geologic disasters and the corresponding mitigation measures (Aurelio 2004). On-site investigations should cover the regional structure, topography, landform, stratigraphic composition, physical and mechanical properties of sediments, hydrodynamic environment, site stability, and geologic hazards (Campbell 1997; Price 2008; Griffiths 2019; Nian et al. 2019; Guo et al. 2020a; Fan et al. 2020b; Shen and Shen 2022; Nian et al. 2022). Based on the above, and with reference to the mature evaluation process for onshore engineering geology environments, which has been implemented for decades, this study identifies a method to evaluate the engineering geology environments of deep seabed mining sites. Figure 15 presents the two aspects of the evaluation process: (1) The quantification and (2) The grading evaluation of the engineering geology environment.

According to the data of the engineering geology environment obtained in the exploration stage (the structure of the mining area, the deep-sea topography and landform, the stratigraphic composition, the physical and mechanical properties of the sediment, the occurrence forms of minerals, and the hydrodynamic environment), the engineering geology environment of the site is evaluated using a combination of qualitative and quantitative methods. The evaluation is typically based on the construction of an evaluation index system, identification of screening evaluation indicators, determination of appropriate evaluation models, and quantitatively grading the evaluation results (Crosta et al. 2006). Because many factors affect the evaluation results, and different factors interact, fuzzy mathematics and artificial intelligence methods are used at this stage (e.g., the gray evaluation model and pattern recognition) to quantitatively evaluate the engineering geology environment (Corominas et al. 2014; Chen et al. 2019). Subsequently, based on the quantitative evaluation results, the conditions of the engineering geology environment, the development level of geologic disasters, and the degree of influence of mining activities on the engineering geology environment are evaluated hierarchically. Table 2 presents the specific evaluation criteria.

To intuitively show the evaluation process of the engineering geology environment of deep seabed mining sites, Weaver et al. (2022) is used as a case study of polymetallic nodule sites, which are most likely to be the first deep-sea minerals that can be commercially exploited. However, the data from Weaver et al. (2022) are not sufficient to support quantitative calculations; thus, this study provides only a preliminary evaluation here. Polymetallic nodules are distributed in abyssal plains with water depths of 4000-6500 m; they are located within 0.3 m of the surface and are fully or partially covered in seafloor sediments (Sharma and Kodagali 1993; Kuhn et al. 2017, 2020). The seabed sediments around polymetallic nodules generally exhibit high porosity, high water content, and low shear strength (Cochonat et al. 1992; Sharma 2001; Sun et al. 2021). These sediments are clearly stratified, and the upper layer is composed of water, sediments, and polymetallic nodules, making the travel of mining vehicles difficult. The underlying layer is composed of sediments and polymetallic nodules that can provide support for mining vehicles (Grupe et al. 2001; Fan et al. 2021). As an abundant area of polymetallic nodules



Fig. 15 Evaluation of the engineering geology environment of a deep seabed mining site

Table 2 Grading evaluation of the engineering geological environments of deep seabed mining sites

Туре	Favorable	Moderate	Complicated
Engineering geological environmental conditions	Simple	Relatively simple	Complicated
Development of geological hazards	No development	Moderate development	Extensive development
Degree of influence of mining activities on the engineer- ing geological environment	Almost no effect	Moderate effect	Serious effect

(Sparenberg 2019; Hein et al. 2020), the Clarion-Clipperton Zone (CC zone) is located south of the Clarion fault zone, east of the Lane fault zone, north of the Clipperton fault zone, and west of the East Pacific Rise, as shown in Fig. 16. There are also secondary fault structures developed between these major structures (Liu et al. 2009; Kuhn et al. 2020). The overall hydrodynamic environment in the area is weak (Aleynik et al. 2017; Fan et al. 2021). Some exploration work carried out in the CC zone (Kotlinski and Stoyanova 2007) has shown that polymetallic nodules are most abundant in depressions between hills and in the plains at the tops of the mounds. The abundance of polymetallic nodules is positively correlated with the slope gradient in the range of 0° to 3.6°. Sampling tests indicated that the penetration resistance of sediments within 20 cm in the surface layer was 0–15 kPa (Liu et al. 2014). Roughly speaking,

the engineering geology environment of this polymetallic nodule site is relatively simple, geologic disasters are moderately developed, the mining activities have had a moderate impact on the engineering geology environment, and the overall grade of the engineering geology environment in the CC zone is moderate.

4.2 Geologic disasters at deep seabed mining sites

Geologic disasters in deep seabed mining sites may be caused by the deterioration of the geologic environment (e.g., the disturbance of sediments brought by mining activities), which changes the physical and mechanical properties of the sediments. In the absence of the influence of mining vehicles, geologic disasters may occur in seabed mining sites due to the instability of the seabed under the action of



Fig. 16 Tectonic belt surrounding the CC zone (modified from Zhang et al. 2001; Li 2008; Liu et al. 2009; Hein 2016; Miller et al. 2018)

external loads. These loads include tectonic action, gravity, hydrodynamic action, gas release, and changes in pore pressure (Song 1993; Rahman 1994; Jeng 2001; Cochonat et al. 2002; Basilone et al. 2014; Zhang et al. 2021b). The influence of tectonic action is closely related to the distribution and occurrence forms of minerals (e.g., polymetallic sulfides), which are formed by cooling hydrothermal eruptions and are widely distributed in tectonically active areas (e.g., mid-ocean ridges, intraplate volcanoes, and post-arc basins) (Herzig and Hannington 1995; Hannington and Monecke 2009). The CC zone is located between the Clarion and Clipperton faults in the North Pacific and is affected by the fault tectonics developed there (Kuhn et al. 2020). Unlike polymetallic sulfides and polymetallic nodules, cobalt-rich crusts are mostly found on the sides and peaks of seamounts, plateaus, and abyssal hills far from tectonically active zones (Keating 1989; Hein et al. 2017) and are generally consolidated on bedrock. As the maturity of the crust increases, the porosity and moisture content decrease, leading to the dehydration of water-rich sediments along with the shrinking, cracking, and slipping of cobalt-rich crusts and bedrock (Keating 1989; Hein et al. 2017). Simultaneously, the crust sites are strongly affected by hydrodynamics (Shi et al. 2021), which exacerbate crustal shedding and result in

Fig. 17 Comparison of geological conditions of different deposit areas of deep seabed minerals (modified from Lusty and Murton 2018)

geologic disasters. Figure 17 compares the geologic conditions of different deep seabed mineral deposit areas.

Mining activities can disturb sediments via the travel of mining vehicles and the dragging of the ground-touching part of the conveying hose. Seabed sediments at polymetallic nodule sites have high fluidization, poor shear resistance, large porosity, and high water content with obvious stratification (Song 1999; Gillard et al. 2019). Under these conditions, seafloor sediments may subside when subjected to gravitational loads from mining vehicles (Grupe et al. 2001; Dai 2010). In addition, the travel of mining vehicles and dragging of the ground-touching part of the conveying hose can compact the sediments at the rut, and the sediments on both sides are turned over before redeposition (Miller et al. 2018). This leads to large changes in the physical and mechanical properties of the sediments (Fan et al. 2021), as shown in Fig. 18.

4.3 Technological advances for the in-situ monitoring of geologic environments

Deep seabed mining is a complex activity that requires the long-term, real-time monitoring of various parameters to ensure sustainable, economic, reliable, and safe operations. However, the complex and dynamic geologic environment requires the appropriate selection of in-situ monitoring techniques to minimize losses during mining activities. Techniques for monitoring the contents of the geologic environment of deep seabed mining can be divided into three categories: the monitoring of external loads, internal

Fig. 18 Compaction zone and redeposition zone of the seabed mining track (modified from Miller et al. 2018; Cuyvers et al. 2018)

changes, and macroscopic deformation of geologic masses. However, few in-situ monitoring techniques have been developed specifically for the geologic environment of deep seabed mining activities. Thus, monitoring techniques for marine geologic disasters are summarized here to encourage the development of new monitoring techniques or apply existing techniques for the long-term or even ultra-long-term monitoring of deep seabed mining activities.

The external loads at deep seabed mining sites can be monitored by evaluating seismic action, tectonic movements, and the external dynamic environment (Jia et al. 2022). Figure 19 presents some technologies for monitoring external loads. Seafloor seismic action and tectonic movements can be monitored using an ocean bottom seismograph (OBS) (Gohl 2003; Jia et al. 2022). Seafloor tectonic movements, plate subduction, and magmatic activity can be reflected by the shallow heat flow and temperature changes with depth, which can be monitored over the long term by seafloor heat flow probes (Davis et al. 2003; Hamamoto et al. 2005). Monitoring the external dynamic environment of deep seabed mining sites refers to monitoring the hydrodynamic environment. Acoustic Doppler velocimetry can be used to measure the 3D instantaneous flow velocity and concentration (Hosseini et al. 2006). Ji et al. (2022) developed an insitu observation system for the bottom boundary layer in the Abyssal Sea. This system can measure dynamic changes in the physical and chemical parameters of the seawater and the seabed boundary layer (e.g., temperature, salinity, pressure, and redox potential).

Monitoring the internal changes in sediments involves monitoring the pore water pressure, water content, density, and temperature. Figure 20 presents some tools for monitoring the internal changes in sediments, which include pore pressure probes and temperature sensors (Jia et al. 2022). Pore pressure can effectively reflect the external loads and geological processes of the sediments, and the in-situ observation of pore pressure plays a pivotal role in the prediction and prevention of geologic disasters (Schultheiss 1990; Liu et al. 2015). Changes in the internal temperatures of sediments can reflect the movement of fluids in pores and can be used to judge the stability of the seabed (Jia et al. 2022). Sun et al. (2022) developed a complex in-situ long-term monitoring system that can monitor certain parameters (e.g., acoustics, resistivity, and excess pore pressure) to provide early warning of submarine geologic disasters.

The monitoring of the macroscopic deformation of geologic bodies focuses on changes in the seafloor topography (Jia et al. 2022). Monitoring the seabed topography is important for determining whether a geologic disaster exists. While multibeam sonar has been applied to monitor

Fig. 19 Systems for the in-situ monitoring of external loads at deep seabed mining sites **a** Ocean bottom seismograph from the University of Hamburg (modified from Gohl 2003) **b** Ocean bottom seismograph from GeoPro (modified from Gohl 2003) **c** Seafloor heat flow probe

(modified from Nagihara et al. 2002) and **d** In-situ observation system for the bottom boundary layer of the Abyssal Sea (modified from Ji et al. 2022)

seabed topography (Mayer 2006), it is difficult to achieve real-time, long-term monitoring using this approach (Zhu 2019). High-precision pressure gauges and three-axis accelerometers enable the long-term, in-situ monitoring of seabed deformation. Currently, microelectromechanical systems (MEMS) accelerometers and inclinometers are widely used to monitor subsidence and landslide deformation (Ge et al. 2022). Zhu (2019) designed a device that uses MEMS sensors and microcontrollers to monitor seafloor topography changes.

5 Environmental effects related to deep seabed mining

5.1 Environmental protection requirements at different mining stages

Potential environmental issues caused by deep seabed mining activities are important reasons for restricting commercial development (Collins et al. 2013; Jones et al. 2020). ISA clearly stipulates that evaluations of environmental

Fig. 20 Tools for the in-situ monitoring of internal changes in geologic bodies **a** Pore pressure probes (Piezometer probes) (modified from Harvey et al. 1997) **b** IFREMER Piezometer V2 (Stegmann et al. 2011) and **c** In-situ long-term monitoring system (Sun et al. 2022)

disturbance and environmental impact reports are necessary for deep seabed mining (Bräger et al. 2020). Therefore, in-depth investigations are required to identify and predict the environmental impacts of mining activities. This section elaborates on the environmental protection requirements at different mining stages.

(1) During the exploration stage, deep seabed mining sites need to be investigated to clarify the range of natural variations in the environmental baseline parameters and distinguish the anthropogenic disturbances from the natural environmental conditions (Clark et al. 2020; CAOE 2021). The evaluation should consider the diversity of deep-sea organisms, particularly sedentary organisms and macrobenthic organisms with poor mobility (Craik 2008; CAOE 2021). According to the results of the environmental investigations, the impacts of mining activities on the environment, society, and economy are predicted using a combination of qualitative and quantitative methods and incorporated in subsequent reports (Perez and Sanchez 2009; Durden et al. 2018).

- (2) During the extraction stage, the environmental considerations focus on the mining process and tailings. The mining sites should be monitored using the latest techniques under the guidance of theory (Weaver et al. 2022) over a long time period to analyze the impacts of mining activities on the environment (ISA 2009, 2010, 2013), ensure that sensitive environmental parameters remain within acceptable ranges, and verify the rationality of the predicted environmental impacts (CAOE 2021).
- (3) During the closure stage, the main task is to restore the environment of the mining site, particularly with respect to the original species (CAOE 2021). Although the specific impacts of mining activities are not fully understood, the following aspects should be considered: the proportion of the area affected by mining activities out of the total area of the water body; the impacts of mining activities on surface and bottom flows in different seasons; the proximity of coastal zones and residential areas to the affected areas; and the fishing potential or value for other commercial activities in the affected areas (Sharma 2015).

5.2 Effects of deep seabed mining on marine environments

Despite the scarcity of available data, some studies (Heffernan 2019; Weaver et al. 2022) have shown that mining activities can have direct, significant, and even devastating impacts on marine ecosystems (Fig. 21). Thiel and Schriever (1990) simulated the travel of mining vehicles on the seabed of the Pacific Ocean and found that a large amount of marine life could be destroyed even when no minerals were collected; the simulated area did not recover until 2015 (e.g., the tracks left by the mining vehicles were visible, and seabed creatures including sponges, soft corals, and anemones had hardly returned) (Heffernan 2019; Vonnahme et al. 2020). The marine environment is evaluated based on the physical, chemical, geological, and biological compositions of the marine ecosystem and water along with the properties of the seabed and sediments (ISA 2009, 2010, 2013). Different stages of deep seabed mining can impact the marine environment.

Seabed mining activities have two direct environmental impacts: harm or death to organisms (Mestre et al. 2017) and far-reaching or even permanent damage to their habitats (Birney et al. 2006; Gollner et al. 2022). Marine life is still abundant at deep seabed mineral sites that receive almost no light and have temperatures close to 0 °C. The travel of mining vehicles can crush the deep-sea organisms (e.g., polychaete worms and tiny sediment-dwelling creatures growing in seafloor sediments at polymetallic nodule sites

Fig. 21 Schematic diagram of the seabed mining process and its interaction with the ecological environment for the three main deposit types (modified from Miller et al. 2018)

and sponges and other filter feeders; Fig. 22) (Heffernan 2019). In addition, the scraping and cutting of the seabed sediments by mining vehicles along with the pumping of bottom water can injure or kill deep-sea fish and zooplankton that have poor mobility and are unable to escape interference (Koschinsky et al. 2018). The light emitted by seabed mining devices can affect bioluminescence, an important ecological function, and damage the eyes of living organisms (Lukoseviciute 2022).

Mining activities also damage the seabed ecosystem (Van Dover et al. 2017); for example, the exploitation of polymetallic sulfides may destroy the unique benthic ecosystem, which is entirely chemically synthesized, and lead to the acidification of the surrounding seawater (Hallgren and Hansson 2021). Moreover, the jets used to loosen the minerals disturb the sediments and form a seafloor sediment plume; these jets may permanently clear minerals and their surrounding sediments, leading to multifaceted impacts on the fragile ecological environment (Weaver et al. 2018; Paulus 2021). Seabed mining activities also have many indirect impacts that require attention (Birney et al. 2006). The operation of mining vehicles and the movement of seabed minerals through pipelines or chain barrels transmit noise in

Fig. 22 Organisms living in areas where deep seabed mineral resources are found (photos taken by ROV KIEL 6000 during cruise M78/2) **a** Sea cucumber in a polymetallic nodule field **b** Deep-sea crab on a seamount **c** A dense population of deep-sea shrimps at a black smoker

all directions. The low-frequency noise can travel a distance of 600 km, which negatively affects the normal life activities of marine organisms and may interfere with the communication and navigation functions of some equipment (Williams et al. 2022; Gross 2022).

5.3 Effects of plumes from deep seabed mining

In deep seabed mining activities, the collection of minerals, the beneficiation on the ship, and the discharge of underwater tailings have serious impacts on the marine environment, especially plumes (Fig. 23) (Muñoz-Royo et al. 2022; Peacock and Ouillon 2022). The mining of 3×10^6 t of polymetallic nodules per year is estimated to result in a total volume of 5×10^9 m³ of resuspended sediment (Sharma

Fig. 23 Schematic diagram of a deep seabed mining system and plume (modified from Muñoz-Royo et al. 2021; photos taken by JAGO team/GEOMAR and QC2000 Nodule Collector developed by SMD)

2005). A high plume concentration can cover and bury the seabed fauna, causing suffocation and death of the organisms (Heffernan 2019). In addition, organisms located in deep-sea areas have adapted to conditions with low turbidity and low food supply; thus, very low plume concentrations can also affect the ecology of the mining site (Weaver et al. 2022).

There can be two stages of plume generation: the seafloor sediment plume discharged during extraction and the midwater plume discharged during beneficiation and the treatment of underwater tailings (Sharma et al. 2017; Muñoz-Royo et al. 2021). During the extraction stage, the travel of mining vehicles and the dragging of the ground-touching part of the pipelines disturb the seabed sediments and form the seafloor sediment plume (Peacock and Ouillon 2022). The composition of the midwater plume is diverse and includes tailings, sediments brought to the production support vessel, and effluent from mineral cleaning. The evolution and fate of the sediment plume are determined by the transport process, which operates on multiple spatiotemporal scales (Peacock and Ouillon 2022). The propagation range of the plume depends on the duration, velocity, volume, concentration, particle size distribution, and emission depth along with the state of the ocean in the discharge zone (Gillard et al. 2019; Fan et al. 2021). The scale of the sediment plume is underpinned by transport phenomena and fundamental fluid mechanics and depends on the design of the mining system and rheology of sediment-laden flows (Guo et al. 2020b, 2021b), the flocculation and settling behavior of the sediments, and the process of ambient flow (Spearman et al. 2020; Peacock and Ouillon 2022). For a 12×12 km² nodular mining site, the diffusion distance of particles larger than 0.1 mm is estimated to reach 50 km in one year (Ding et al. 2021). The most direct way to reduce the environmental impact of plumes is to release midwater plumes as close as possible to the seafloor (Muñoz-Royo et al. 2021) or in the low disphotic or aphotic zone; this reduces the vertical migration of plume material to minimize the impact on photosynthesis (Muñoz-Royo et al. 2021). ISA intends to introduce regulations to control the impacts of plumes on the deep-sea environment (Weaver et al. 2022). While the extent and specific impact of plumes have received considerable attention, simulating and predicting the propagation of large-scale plumes remain important issues to be addressed (Fan et al. 2021; Muñoz-Royo et al. 2021). Studies involving physical model tests, long-term in-situ monitoring, and advanced numerical simulation are needed to reveal the migration mechanism of plumes and assess the spatial extent of their impact.

6 Summary and conclusions

The commercial exploitation of deep seabed mineral resources is gradually approaching reality, and efforts are underway to ensure that deep seabed mining activities are sustainable, economic, efficient, and safe. In this study, the development of ocean mineral resources, the techniques used for ocean mining, the engineering geology issues related to deep seabed mining, and the environmental effects of deep seabed mining are analyzed. A brief summary and key conclusions are presented as follows.

(1) A bibliometric analysis was conducted on the development of ocean mineral resources based on 5838 studies from the WOS and CNKI databases. The numbers of relevant studies in this field gradually increased from 1957 to 2022, and the development trend can be roughly divided into three stages: Stage I (1957 to 1988), Stage II (1989 to 2008), and Stage III (2009 to 2022). Based on this trend, the topic "ocean mineral resources" has been a research hotspot in more than past 10 years and will continue to be for a long time in the future. Moreover, the continent with the most research in this area is Europe, and international cooperation on this research topic is frequent. A cluster analysis on the keywords of the research studies indi-

cated that the most popular research directions include mineral resource characteristics, resource development approaches, and environmental effects.

- (2) According to the three stages of ocean mining activities (i.e., exploration, extraction, and closure), the development process, theory, and issues related to the main techniques involved in each stage were introduced. In the exploration stage, ocean mineral resources are mainly found based on differences in geophysical field features (e.g., density, magnetic permeability, electrical conductivity, and radioactivity) or by geotechnical testing using in-situ and laboratory techniques. These geophysical and geotechnical exploration techniques are usually used simultaneously. The techniques applied in the extraction stage can generally be divided into mobile and fixed extraction techniques. Subsea mining vehicles, which provide a high level of control and high mining efficiency, are the latest tool developed for the mobile extraction of deep-sea resources. Fixed extraction in deep-sea environments is primarily conducted using semi-submersible platforms that can be used to extract multiple resources (e.g., oil, natural gas, combustible ice, brimstone, and coal). Finally, techniques applied during closure are still in the exploratory stage and include the construction of artificial algae farms, the planting of mangroves, and the placement of artificial reefs. However, these techniques are receiving extensive research attention and show good prospects for the future.
- (3) Based on data from an in-situ engineering geology investigation, a method for evaluating the engineering geology environment of a deep seabed mining site was proposed. The CC zone located in the Eastern Pacific Ocean, which is surrounded by tectonic belts, was used as a case study, and data were obtained from the published literature. The overall hydrodynamic environment in this area is weak, the penetration resistance of sediments within 20 cm of the surface layer is 0–15 kPa, the engineering geology environment is relatively simple, geologic disasters are moderately developed, and the impact of mining activities on the engineering geology environment is moderate; thus, the overall grade of the engineering geology environment is moderate. Geologic disasters at deep seabed mining sites may be caused by the disturbance of sediments by mining activities and the instability of the seabed under the action of external loads. Accordingly, techniques for the in-situ monitoring of the geologic environment are required to prevent and/or minimize losses. The different monitoring objects can be divided into three categories: external loads, internal changes, and macroscopic deformation of geologic masses. Various corresponding monitoring techniques and tools have been

developed, including OBS, seafloor heat flow probes, acoustic Doppler velocimetry, and pore pressure probes. However, few in-situ monitoring techniques have been developed specifically for the geologic environments of deep seabed mining sites. Thus, the development of monitoring techniques or application of existing monitoring techniques for the long-term and ultra-long-term monitoring of deep seabed mining sites is encouraged.

The environmental protection requirements during the (4) three stages of deep seabed mining and the environmental effects of deep seabed mining were elucidated through an investigation of existing regulations and literature. Each mining stage involves different environmental considerations for the mining process, tailings treatment, and the restoration of the environment. Deep seabed mining activities have both direct and indirect impacts on the marine environment, including damage to deep-sea organisms and their habitats, noise transmission in seawater, and the formation of sediment plumes. Predicting the migration mechanism and controlling the propagation of large-scale plumes remain important problems to be solved through physical experiments, numerical simulation, and in-situ monitoring.

Acknowledgements The authors thank all the scholars responsible for the referenced work and professional websites for providing figures.

Authors' contributions XG: Conceptualization, Methodology, Investigation, Visualization, Writing—Original Draft. NF: Methodology, Investigation, Visualization, Writing—Original Draft. YL: Investigation, Visualization, Writing—Original Draft. XL: Project administration, Methodology, Resources, Supervision, Writing—Review & Editing. ZW: Visualization, Writing—Review & Editing. XX: Visualization, Writing—Review & Editing. YJ: Project administration, Writing—Review & Editing. XG and NF contributed equally to this work.

Funding Funding for this research was provided by the National Natural Science Foundation of China (42022052, 42277138, and 52108337), the National Key R&D Program of China (2022YFC2803800), the Shandong Provincial Natural Science Foundation (ZR2020YQ29), UCL's Department of Civil, Environmental and Geomatic Engineering, and Ocean University of China. This support is gratefully acknowledged.

Availability of data and materials All data, models, and code generated or used during this study appear in the submitted article.

Declarations

Competing interests All the authors of this manuscript have approved the submission of this article for publication, and there are no conflicts of interest to declare. This paper has not been published elsewhere and is not under consideration by another journal.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing,

adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Aitcin PC, Houle J (1986) Design and construction of concrete sea structures. Can J Civil Eng 13(4):498–499. https://doi.org/10. 1139/L86-074
- Aleynik D, Inall ME, Dale A, Vink A (2017) Impact of remotely generated eddies on plume dispersion at abyssal mining sites in the Pacific. Sci Rep 7(1):1–14. https://doi.org/10.1038/ s41598-017-16912-2
- Allaud L, Martin MJ (1977) Schlumberger: the history of a technique, John Wiley & Sons Inc. https://doi.org/10.1017/S001675680 0042102
- American Bureau of Shipping (2020) Guide for subsea mining. https:// ww2.eagle.org/content/dam/eagle/rules-and-guides/current/offsh ore/318 guideforsubseamining/subsea-mining-guide-oct20.pdf
- Aurelio MA (2004) Engineering geological and geohazard assessment (EGGA) system for sustainable infrastructure development: the Philippine experience. Bull Geol Soc Hong Kong 7:3339
- Ballu VS, Hildebrand JA, Webb SC (1998) Seafloor gravity evidence for hydrothermal alteration of the sediments in middle valley. Juan De Fuca Ridge Mar Geol 150:99–111. https://doi.org/10. 1016/S0025-3227(98)00024-3
- Basilone L, Lena G, Gasparo-Morticelli M (2014) Synsedimentarytectonic, soft-sediment deformation and volcanism in the rifted Tethyan margin from the Upper Triassic-Middle Jurassic deepwater carbonates in Central Sicily. Sed Geol 308:63–79. https:// doi.org/10.1016/j.sedgeo.2014.05.002
- Bath AR (1989) Deep sea mining technology: recent developments and future projects. Offshore Technol Conf. https://doi.org/10. 4043/5998-MS
- Begemann HKSP (1953) Improved method of determining resistance to adhesion by sounding through a loose sleeve placed behind the cone. In: The 3rd International conference on soil mechanics and foundation engineering. https://www. issmge.org/publications/publication/improved-methodof-determining-resistance-to-adhesion-by-sounding-throu gh-a-loose-sleeve-placed-behind-the-cone
- Birney K, Griffin A, Gwiazda J, Kefauver J, Nagai T, Varchol D (2006) Potential deep-sea mining of seafloor massive sulfides: a case study in Papua New Guinea. Donald Bren School of Environmental Science and Management Thesis. https://actnowpng.org/ sites/default/files/ventsthesis.pdf
- Bräger S, Rodriguez GQR, Mulsow S (2020) The current status of environmental requirements for deep seabed mining issued by the International Seabed Authority. Mar Policy 114:103258. https:// doi.org/10.1016/j.marpol.2018.09.003
- Briaud JL (2013) In situ tests. Wiley. https://doi.org/10.1002/97811 18686195.ch7
- Cadling L, Odenstad S (1950) Vane borer. An apparatus for determining the shear strength of clay soils directly in the ground. https:// www.sgi.se/en/research--learning/publikationer/sgi-proceedings/

vane-borer.-an-apparatus-for-determining-the-shear-strength-ofclay-soils-directly-in-the-ground/

- Campbell KJ (1997) Deepwater geohazards and engineering geology: meeting tough challenges. https://archives.datapages.com/data/ HGS/vol40/no01/09.htm
- Chen C (2004) Searching for intellectual turning points: Progressive knowledge domain visualization. Proc Natl Acad Sci U S A 101(Suppl. 1):5303–5310. https://doi.org/10.1073/pnas.03075 13100
- Chen C (2006) CiteSpace II: detecting and visualizing emerging trends and transient patterns in scientific literature. J Am Soc Inf Sci Technol 57(3):359–377. https://doi.org/10.1002/asi.20317
- Chen X (2015) The application and research of magnetic exploration in exploration of iron mining area in Wenchuan. Chengdu University of Technology, Chengdu (**in Chinese**). https://cdmd.cnki. com.cn/Article/CDMD-10616-1015312565.htm
- Chen W, Shahabi H, Shirzadi A, Hong H, Akgun A, Tian Y, Liu J, Zhu A, Li S (2019) Novel hybrid artificial intelligence approach of bivariate statistical-methods-based kernel logistic regression classifier for landslide susceptibility modeling. Bull Eng Geol Env 78(6):4397–4419. https://doi.org/10.1007/ s10064-018-1401-8
- China Association of Oceanic Engineering (2021) Guidelines for environment protection and preservation of deep-sea mining activities, T/CAOE 41–2021a (**in Chinese**). http://caoe.org.cn/upload/ 202112/2021121014431611269540.pdf
- Chung JS (1998) An articulated pipe-miner system with thrust control for deep-ocean crust mining. Mar Georesour Geotechnol 16(4):253–271. https://doi.org/10.1080/10641199809379971
- Chung JS (2003) Deep-ocean technology: learning curve. In: Proceedings of 5th Ocean mining symposium, Tsukuba, Japan. https:// www.researchgate.net/publication/290362656_Deep-ocean_mining_technology_Learning_curve_I
- Clark MR, Durden JM, Christiansen S (2020) Environmental impact assessments for deep-sea mining: can we improve their future effectiveness? Mar Policy. https://doi.org/10.1016/j.marpol.2018. 11.026
- Cochonat P, Le Suavé R, Charles C, Greger B, Hoffert M, Lenoble JP, Meunier J, Pautot G (1992) First in situ studies of nodule distribution and geotechnical measurements of associated deep-sea clay (Northeastern Pacific Ocean). Mar Geol 103(1–3):373–380. https://doi.org/10.1016/0025-3227(92)90027-F
- Cochonat P, Cadet JP, Lallemant SJ, Mazzotti S, Nouzé H, Fouchet C, Foucher JP (2002) Slope instabilities and gravity processes in fluid migration and tectonically active environment in the eastern Nankai accretionary wedge (KAIKO-Tokai'96 cruise). Mar Geol 187(1–2):193–202. https://doi.org/10.1016/S0025-3227(02) 00266-9
- Collins PC, Kennedy B, Copley J, Boschen R, Fleming N, Forde J, Ju S, Lindsay D, Marsh L, Nye V, Patterson A, Watanabe H, Yamamoto H, Carlsso J, Thaler AD (2013) VentBase: developing a consensus among stakeholders in the deep-sea regarding environmental impact assessment for deep-sea mining—a workshop report. Mar Policy 42:334–336. https://doi.org/10.1016/j. marpol.2013.03.002
- Constable S (2010) Ten years of marine CSEM for hydrocarbon exploration. Geophysics 75(5):7567–7581. https://doi.org/10.1190/1. 3483451
- Corominas J, Van Westen C, Frattini P, Cascini L, Malet JP, Fotopoulou S, Catani F, Van Den Eeckhaut M, Mavrouli O, Agliardi F, Pitilakis K, Winter MG, Pastor M, Ferlisi S, Tofani V, Hervás J, Smith JT (2014) Recommendations for the quantitative analysis of landslide risk. Bull Eng Geol Env 73(2):209–263. https://doi. org/10.1007/s10064-013-0538-8

- Corwin RF (1976) Offshore use of the self-potential method. Geophys Prospect 24(1):79–90. https://doi.org/10.1111/j.1365-2478.1976. tb00386.x
- Craik N (2008) The international law of environmental impact assessment: process, substance and integration. Cambridge University Press. https://doi.org/10.1017/CBO9780511494611
- Crosta GB, Chen H, Frattini P (2006) Forecasting hazard scenarios and implications for the evaluation of countermeasure efficiency for large debris avalanches. Eng Geol 83(1–3):236–253. https://doi. org/10.1016/j.enggeo.2005.06.039
- Cuyvers L, Berry W, Gjerde K, Thiele T, Wilhem C (2018) Deep seabed mining, a rising environmental challenge. IUCN, Gland. https://doi.org/10.2305/IUCN.CH.2018.16.en
- Dai Y (2010) The modelling research and simulation analysis on the single-rigid-body of tracked miner moving on the seafloor. Central South University (in Chinese). https://cdmd.cnki.com.cn/ article/cdmd-10533-2010185642.htm
- Davis EE, Wang K, Becker K, Thomson RE, Yashayaev I (2003) Deep-ocean temperature variations and implications for errors in seafloor heat flow determinations. J Geophys Res Solid Earth 108:11. https://doi.org/10.1029/2001JB001695
- Dayal U, Allen JH (1975) The effect of penetration rate on the strength of remolded clay and sand samples. Can Geotech J 12(12):336–348. https://doi.org/10.1139/t75-038
- Ding L, Chen X, Gao Y (2009) Seafloor massive sulfides: the frontier of deep ocean miming. Ocean Technol 28(1):126–132
- Ding ZJ, Sun YF, Gao W, Ren YG (2021) Study on potential environmental impact factors and monitoring technology system of deep-sea mining. Ocean Dev Manag 38(7):73–83. http://www. haiyangkaifayuguanli.com/hykfyglen/ch/reader/view_abstract. aspx?file_no=20210712&flag=1
- Dong Y, Cui L, Zhang X (2022) Multiple-GPU parallelization of three-dimensional material point method based on single-root complex. Int J Numer Methods Eng 123(6):1481–1504. https:// doi.org/10.1002/nme.6906
- Dong Y, Liao Z, Wang J, Liu Q, Cui L (2023) Potential failure patterns of a large landslide complex in the Three Gorges Reservoir area. Bull Eng Geol Env 82(1):1–20. https://doi.org/10. 1007/s10064-022-03062-7
- Du XG, Guan LQ, Zhuo WX (2016) Analysis of the current situation of deep sea mining development and demand for deep sea mining vessels in China. Strait Sci 12:62–67
- Durden JM, Lallier LE, Murphy K, Jaeckel A, Gjerde K, Jones DO (2018) Environmental impact assessment process for deep-sea mining in 'the Area.' Mar Policy 87:194–202. https://doi.org/ 10.1016/j.marpol.2017.10.013
- Essa KS, Mehanee SA, Soliman KS, Diab ZE (2020) Gravity profile interpretation using the R-parameter imaging technique with application to ore exploration. Ore Geol Rev 126:103695. https://doi.org/10.1016/j.oregeorev.2020.103695
- Evans RL (2007) Using CSEM techniques to map the shallow section of seafloor: from the coastline to the edges of the continental slope. Geophysics 72:105–116. https://doi.org/10.1190/1. 2434798
- Evans RL, Law LK, Louis BS, Cheesman S, Sananikone K (1999) The shallow porosity structure of the continental shelf of the Eel Shelf, northern California: results of a towed electromagnetic survey. Mar Geol 154(1–4):211–226. https://doi.org/10. 1016/S0025-3227(98)00114-5
- Fan N, Nian TK, Guo XS, Jiao HB, Lu S (2020a) Piecewise strength model for three types of ultra-soft fine-grained soils. Soils Found 60(4):778–790. https://doi.org/10.1016/j.sandf.2020. 04.010
- Fan N, Nian TK, Jiao HB, Guo XS, Zheng DF (2020b) Evaluation of the mass transfer flux at interfaces between submarine sliding

soils and ambient water. Ocean Eng 216:108069. https://doi.org/ 10.1016/j.oceaneng.2020.108069

- Fan Z, Jia Y, Teng X, Sun Z, Song X, Yang G, Wang L, Li B (2021) Review on potential engineering geological environment impacts of deep-sea polymetallic nodules mining. J Eng Geol 29(6):1676–1691. https://doi.org/10.13544/j.cnki.jeg.2021-0260
- Fan N, Jiang J, Dong Y, Guo L, Song L (2022) Approach for evaluating instantaneous impact forces during submarine slidepipeline interaction considering the inertial action. Ocean Eng 245:110466. https://doi.org/10.1016/j.oceaneng.2021.110466
- Fell R, Corominas J, Bonnard C, Cascini L, Leroi E, Savage WZ (2008) Guidelines for landslide susceptibility, hazard and risk zoning for land use planning. Eng Geol 102(3–4):85–98. https://doi.org/10. 1016/j.enggeo.2008.03.022
- Fisher R, O'Leary RA, Low-Choy S, Mengersen K, Knowlton N, Brainard RE, Caley MJ (2015) Species richness on coral reefs and the pursuit of convergent global estimates. Curr Biol 25(4):500–505. https://doi.org/10.1016/j.cub.2014.12.022
- Forsberg R, Olesen AV, Einarsson I (2015) Airborne gravimetry for geoid determination with Lacoste Romberg and Chekan gravimeters. Gyroscopy Navig 6(4):265–270. https://doi.org/10.1134/ S2075108715040069
- Fujimoto H, Koizumi K, Watanabe M, Oshida A, Furuta T, Takamura N, Ura T (2000) Underwater gravimeter on board the R-One Robot. In: The 2000a international symposium on underwater technology. IEEE Catalog No.00EX418. https://doi.org/10.1109/ UT.2000.852559
- Gauthier M, Charles C (1979) French technological development in nodule mining. In: Offshore technology conference. OnePetro. https://doi.org/10.4043/3369-MS
- Ge Y, Chen J, Zhang P, Cao C, Le X, Ai J, Zhou P, Liang T (2022) A novel technique for seabed strata deformation in-situ monitoring. Front Mar Sci. https://doi.org/10.3389/fmars.2022.987319
- Gilbert LA, McDuff RE, Johnson HP (2007) Porosity of the upper edifice of axial seamount. Geology 35:49–52. https://doi.org/10. 1130/G22892A.1
- Gillard B, Purkiani K, Chatzievangelou D, Vink A, Iversen MH, Thomsen L (2019) Physical and hydrodynamic properties of deep sea mining-generated, abyssal sediment plumes in the Clarion Clipperton Fracture Zone (eastern-central Pacific). Elem Sci Anthrop. https://doi.org/10.1525/elementa.343
- Gohl K (2003) Structure and dynamics of a submarine continent: tectonic-magmatic evolution of the Campbell Plateau (New Zealand)-Report of the RV SONNE cruise SO-169, Projekt CAMP, 17 January to 24 February 2003b. Bremerhaven: Alfred-Wegener-Institut für Polar-und Meeresforschungpp. (Berichte zur Polar-und Meeresforschung; 457), 88. https://doi.org/10.2312/ BzPM_0457_2003
- Gollner S, Haeckel M, Janssen F, Lefaible N, Molari M, Papadopoulou S, Reichart GJ, Alexandre JT, Vink A, Vanreusel A (2022) Restoration experiments in polymetallic nodule areas. Integr Environ Assess Manag 18(3):682–696. https://doi.org/10.1002/ieam.4541
- Goto TN, Kasaya T, Machfiyama H, Takagi R, Matsumoto R, Okuda Y, Satoh M, Watanabe T, Seama N, Mikada H (2008) A marine deep-towed DC resistivity survey in a methane hydrale area. Japan Sea Explor Gcophy. https://doi.org/10.1071/EG08003
- Graziano F, Anita M (2005) Robert mallet and the 'Great Neapolitan Earthquake' of 1857. Notes Rec Roy Soc. https://doi.org/10. 1098/rsnr.2004.0076
- Grebe H (1997) General mathematical model for string connections between mobile deep-sea devices and their mother stations. Siegen Universitat-GH Siegen. https://www.amazon.de/-/en/Henni ng-Grebe/dp/382652778X
- Griffiths JS (2019) Advances in engineering geology in the UK 1950– 2018. Q J Eng Geol Hydrogeol 52(4):401–413. https://doi.org/ 10.1144/qjegh2018-171

- Gross M (2022) Mining noise set to rock the oceans. https://doi.org/ 10.1016/j.cub.2022.07.046
- Grupe B, Becker HJ, Oebius HU (2001) Geotechnical and sedimentological investigations of deep-sea sediments from a manganese nodule field of the Peru Basin. Deep Sea Res Part II 48(17– 18):3593–3608. https://doi.org/10.1016/S0967-0645(01)00058-3
- Guo XS, Nian TK, Wang ZT, Zhao W, Fan N, Jiao HB (2020a) Lowtemperature rheological behavior of submarine mudflows. J Waterw Port Coast Ocean Eng 146(2):04019043. https://doi. org/10.1061/(ASCE)WW.1943-5460.0000551
- Guo XS, Zheng DF, Nian TK, Lv LT (2020b) Large-scale seafloor stability evaluation of the northern continental slope of South China Sea. Mar Georesour Geotechnol 38(7):804–817. https:// doi.org/10.1080/1064119X.2019.1632996
- Guo XS, Nian TK, Gu ZD (2021a) A fluid mechanics approach to evaluating marine soft clay strength by a ball full-flow penetrometer. Appl Ocean Res 116:102865. https://doi.org/10.1016/j.apor. 2021.102865
- Guo XS, Nian TK, Gu ZD, Li DY, Fan N, Zheng DF (2021b) Evaluation methodology of laminar-turbulent flow state for fluidized material with special reference to submarine landslide. J Waterw Port Coast Ocean Eng 147(1):04020048. https://doi. org/10.1061/(ASCE)WW.1943-5460.0000616
- Guo XS, Nian TK, Wang D, Gu ZD (2022a) Evaluation of undrained shear strength of surficial marine clays using ball penetrationbased CFD modelling. Acta Geotech 17(5):1627–1643. https:// doi.org/10.1007/s11440-021-01347-x
- Guo XS, Nian TK, Zhao W, Gu ZD, Liu CP, Liu XL, Jia YG (2022b) Centrifuge experiment on the penetration test for evaluating undrained strength of deep-sea surface soils. Int J Min Sci Technol 32(2):363–373. https://doi.org/10.1016/j.ijmst.2021. 12.005
- Gu Z, Guo X, Nian T, Fu C, Zhao W (2022) An improved evaluation method for the undrained shear strength of uniform soft clay in the nonfull flow state based on ball penetration simulations. Appl Ocean Res 128:103365. https://doi.org/10.1016/j.apor. 2022.103365
- Guo XS, Nian TK, Fu CW, Zheng DF (2023a) Numerical investigation of the landslide cover thickness effect on the drag forces acting on submarine pipelines. J Waterw Port Coast Ocean Eng 149(2):04022032. https://doi.org/10.1061/JWPED5. WWENG-1869
- Guo XS, Stoesser T, Zheng DF, Luo QY, Liu XL, Nian TK (2023b) A methodology to predict the run-out distance of submarine landslides. Comput Geotech 153:105073. https://doi.org/10.1016/j. compgeo.2022.105073
- Guo X, Liu Z, Zheng J, Luo Q, Liu X (2023c) Bearing capacity factors of T-bar from surficial to stable penetration into deep-sea sediments. Soil Dyn Earthq Eng 165:107671. https://doi.org/10. 1016/j.soildyn.2022.107671
- Guzzetti F, Reichenbach P, Cardinali M, Galli M, Ardizzone F (2005) Probabilistic landslide hazard assessment at the basin scale. Geomorphology 72(1–4):272–299. https://doi.org/10.1016/j.geomo rph.2005.06.002
- Hallgren A, Hansson A (2021) Conflicting narratives of deep sea mining. Sustainability 13(9):5261. https://doi.org/10.3390/su130 95261
- Hamamoto H, Yamano M, Goto S (2005) Heat flow measurement in shallow seas through long-term temperature monitoring. Geophys Res Lett. https://doi.org/10.1029/2005GL024138
- Hannington M, Monecke T (2009) Global exploration models for polymetallic sulphides in the Area: an assessment of lease block selection under the draft regulations on prospecting and exploration for polymetallic sulphides. Mar Georesour Geotechnol 27(2):132–159. https://doi.org/10.1080/10641190802680614

- Hanzawa H, Tanaka H (1992) Normalized undrained strength of clay in the normally consolidated state and in the field. Soils Found 32(1):132–148. https://doi.org/10.3208/sandf1972.32.132
- Harvey FE, Rudolph DL, Frape SK (1997) Measurement of hydraulic properties in deep lake sediments using a tethered pore pressure probe: applications in the Hamilton Harbour, western Lake Ontario. Water Resour Res 33(8):1917–1928. https://doi.org/10. 1029/97WR00659
- He ZY, Lin JG, Yang BH (2016) The progress and viewpoints on the development of the regulations for mineral exploitation in the area. Pacific J 24(10):9–17. https://doi.org/10.14015/j.cnki.1004-8049.2016.10.002. (in Chinese)
- Heffernan OLIVE (2019) Deep-sea dilemma. Nature 571(7766):465– 468. https://www.nature.com/articles/d41586-019-02242-y
- Hein JR (2016) Manganese nodules. Encyclopedia Mar Geosci 408– 412. https://www.researchgate.net/publication/306107551_ Manganese_Nodules
- Hein JR, Koschinsk A, Bau M, Manheim FT, Kang JK, Roberts L (2017) Cobalt-rich ferromanganese crusts in the Pacific. In Handbook of marine mineral deposits. Routledge, pp 239–279. https:// doi.org/10.1201/9780203752760
- Hein JR, Koschinsky A, Kuhn T (2020) Deep-ocean polymetallic nodules as a resource for critical materials. Nat Rev Earth Environ 1(3):158–169. https://doi.org/10.1038/s43017-020-0027-0
- Herzig PM, Hannington MD (1995) Polymetallic massive sulfides at the modern seafloor a review. Ore Geol Rev 10(2):95–115. https://doi.org/10.1016/0169-1368(95)00009-7
- Hong S, Kim HW, Yeu T, Choi JS, Lee TH, Lee JK (2019) Technologies for safe and sustainable mining of deep-seabed minerals. In: Environmental issues of deep-sea mining: impacts, consequences and policy perspectives. Springer, Cham, pp 95–14. https://doi. org/10.1007/978-3-030-12696-4_5
- Hosseini SA, Shamsai A, Ataie-Ashtiani B (2006) Synchronous measurements of the velocity and concentration in low density turbidity currents using an Acoustic Doppler Velocimeter. Flow Meas Instrum 17(1):59–68. https://doi.org/10.1016/j.flowmeasinst. 2005.05.002
- Howland-Rose AW, Linford G, Pitcher DH, Seigel HO (1980) Some recent magnetic induced-polarization developments—part II: survey results. Geophysics 45(1):55–74. https://doi.org/10. 1190/1.1441040
- Hsu SK, Chiang CW, Evans RL, Chen CS, Chiu SD, Ma YF, Wang Y (2014) Marine controlled source electromagnetic method used for the gas hydrate investigation in the offshore area of SW Taiwan. J Asian Earth Sci 92:224–232. https://doi.org/10.1016/j. jseaes.2013.12.001
- International Seabed Authority (2009) Regulations on prospecting and exploration for cobalt-rich ferromanganese crusts in the Area. ISBA/16/C/WP.2. https://digitallibrary.un.org/record/733356? ln=zh_CN
- International Seabed Authority (2010) Draft regulations on prospecting and exploration for polymetallic sulphides in the Area, ISBA/16/ C/L.5. https://digitallibrary.un.org/record/733335?ln=zh_CN
- International Seabed Authority (2013) Regulations on prospecting and exploration for polymetallic nodules in the area, ISBA/19/C/ WP.1. https://digitallibrary.un.org/record/766000?ln=zh_CN
- James CK, Maurice AT, Dana RY (2013) Dynamics and navigation of autonomous underwater vehicles for submarine gravity surveying. Geophysics 78(3):55–68. https://doi.org/10.1190/geo20 12-0181.1
- Jeng DS (2001) Mechanism of the wave-induced seabed instability in the vicinity of a breakwater: a review. Ocean Eng 28(5):537–570. https://doi.org/10.1016/S0029-8018(00)00013-5
- Ji CS, Jia YG, Zhu JJ (2022) R&D and application of the abyssal bottom boundary layer observation system (ABLOS). Earth Sci Front 29(5):265–274. https://doi.org/10.13745/j.esf.sf.2021.9.28

- Jia Y, Chen T, Li P, Li Z, Hu C, Liu X, Shan H (2022) Research progress on the in-situ monitoring technologies of marine geohazards. Chin J Geological Hazard Control 33(3):1–14. https://doi. org/10.16031/j.cnki.issn.1003-8035.2022.03-0
- Jocobson RS, Dorman LM, Purdy GM, Schultz A, Solomon SC (1991) Ocean bottom seismometer facilities available. EOS Trans Am Geophys Union 72(46):506–515. https://doi.org/10.1029/90EO00366
- Jones DO, Ardron JA, Colaço A, Durden JM (2020) Environmental considerations for impact and preservation reference zones for deep-sea polymetallic nodule mining. Mar Policy. https://doi.org/ 10.1016/j.marpol.2018.10.025
- Ke L, Yu KSH, Wong YS, Tam NFY (2005) Spatial and vertical distribution of polycyclic aromatic hydrocarbons in mangrove sediments. Sci Total Environ 340(1–3):177–187. https://doi.org/10. 1016/j.scitotenv.2004.08.015
- Keating SH (1989) Morphology of central pacific seamounts: implications for Mn-crust mining. In Offshore technology conference. OnePetro. https://doi.org/10.4043/6061-MS
- Kim B, Jeong S, Bang E, Shin S, Cho S (2021) Investigation of iron ore mineral distribution using aero-magnetic exploration techniques: case study at Pocheon. Korea Miner 11(7):665. https://doi.org/ 10.3390/min11070665
- Koschinsky A, Heinrich L, Boehnke K, Cohrs JC, Markus T, Shani M, Singh T, Stegen K, Werner W (2018) Deep-sea mining: interdisciplinary research on potential environmental, legal, economic, and societal implications. Integr Environ Assess Manag 14(6):672–691. https://doi.org/10.1002/ieam.4071
- Kotlinski R, Stoyanova V (2007) Buried and surface polymetallic nodule distribution in the eastern clarion–clipperton zone: main distinctions and similarities. In: Advances in geosciences, vol 9: Solid Earth (SE), Ocean Science (OS) & Atmospheric Science (AS), pp 67–74. https://doi.org/10.1142/9789812708946_0006
- Kuhn T, Wegorzewski A, Rühlemann C, Vink A (2017) Composition, formation, and occurrence of polymetallic nodules. In: Deep-sea mining. Springer, Cham, pp 23–63. https://www.worldscientific. com/doi/abs/10.1142/9789812708946_0006
- Kuhn T, Uhlenkott K, Vink A, Rühlemann C, Arbizu PM (2020) Manganese nodule fields from the Northeast Pacific as benthic habitats. In: Seafloor geomorphology as benthic habitat. Elsevier, pp 933–947. https://doi.org/10.1016/B978-0-12-814960-7.00058-0
- Lenoble JP (1981) Polymetallic nodules resources and reserves in the north pacific from the data collected by Afernod. Ocean Manag 7(1–4):9–24. https://doi.org/10.1016/0302-184X(81)90003-2
- Li GS (2008) Geological characteristics and forming environment of the polymetal nodules in the eastern pacific CC area. Geol Resour 17(4):286–291 (in Chinese). https://www.cgsjournals. com/article/doi/10.13686/j.cnki.dzyzy.2008.04.001
- Lian X, Hu H, Li T, Hu D (2020) Main geological and mining factors affecting ground cracks induced by underground coal mining in Shanxi Province, China. Int J Coal Sci Technol 7(2):362–370. https://doi.org/10.1007/s40789-020-00308-1
- Liu YG, Du DW, Li ZS, Yan SJ, Yang G, Zhu ZW, Wang CJ (2009) Estimation of polymetallic nodule distributionand resource quantity in the CC zone and its adjacent areas of the pacific ocean. Advances in Marine Science 27(3):342–350 (**in Chinese**). https://en.cnki.com.cn/Article_en/CJFDTOTAL-HBHH200903 006.htm
- Liu SJ, Liu C, Dai Y (2014) Status and progress on researches and developments of deep ocean mining equipments. J Mech Eng 50(2):8–18. https://doi.org/10.3901/JME.2014.02.008.
- Liu T, Cui F, Zhang MX (2015) Research progress of deep seabed pore pressure observation techniques in situ. J Hydraul Eng 46(S1):111–116 (in Chinese). https://www.researchgate.net/ publication/283026332_Research_progress_of_deep_seabed_ pore_pressure_observation_techniqu

- Liu XY, Cai GJ, Liu LL, Liu SY, Duan WH, Puppala AJ (2021a) Improved p-y curve models for large diameter and super-long cast-in-place piles using piezocone penetration test data. Comput Geotech 130:103911. https://doi.org/10.1016/j.compgeo.2020. 103911
- Liu Y, Li JB, Wang YJ, Tao CH, Zeng ZG, Han XQ, Li CS, Zhou YD, Liang J (2021b) Development and prospect of exploration and research on seafloor polymetallic sulfides of China. Chin J Nonferrous Met 31(10):2624–2637. https://doi.org/10.11817/j. ysxb.1004.0609.2021-37992
- Liu DH, Wan L, Wang CJ, Li CL (2022) Environmental impact analysis and management countermeasures based on the whole process of deep-sea mining. Adv Mar Sci 40(3):367–378. https://doi.org/ 10.12362/j.issn.1671-6647.20211014001
- Low HE, Randolph MF, Rutherford C, Bernie BB, Brooks JM (2008) Characterization of near seabed surface sediment. In: Proceedings of Offshore Technology conference, Offshore Technology, Houston, Paper OTC 19149. https://doi.org/10.4043/19149-MS
- Lukoseviciute SM (2022) Framework for environmental risk assessment and management of deep-sea mineral mining of massive seafloor sulfide deposits. Master's thesis, uis. https://uis.brage. unit.no/uis-xmlui/bitstream/handle/11250/3026211/no.uis:inspe ra:108218473:21488875.pdf?sequence=1
- Lunne T, Robertsen PK, Powell J (2009) Cone penetration testing in geotechnical practice. Soil Mech Found Eng 46:237. https://doi. org/10.1007/s11204-010-9072-x
- Luo YC (2009) Improve the method theory and the applied research of seismic survey resolution—deconvolution. Chengdu University of Technology, Chengdu (in Chinese). https://cdmd.cnki.com. cn/Article/CDMD-10616-2009221416.htm
- Lusty PA, Murton BJ (2018) Deep-ocean mineral deposits: metal resources and windows into earth processes. Elem Int Mag Miner Geochem Petrol 14(5):301–306. https://doi.org/10.2138/ gselements.14.5.301
- Lykousis V, Sakellariou D, Locat J (2007) Submarine mass movements and their consequences: 3rd international symposium. Kluwer Academic Publishers, Amsterdam. https://link.springer.com/ content/pdf/bfm:978-1-4020-6512-5/1.pdf
- Masuda Y, Cruickshank MJ, Mero JL (1971) Continuous bucket-line dredging at 12000 feet. In: Offshore technology conference, Houston, Texas, pp 94–99. https://doi.org/10.4043/1410-MS
- Mayer LA (2006) Frontiers in seafloor mapping and visualization. Mar Geophys Res 27(1):7–17. https://doi.org/10.1007/ s11001-005-0267-x
- Mayne PW, Coop MR, Springman SM, Huang A, Zornberg JG (2009) Geomaterial behavior and testing. In: Presented at the 17th international conference on soil mechanics and geotechnical engineering, Alexandria, Egypt, vol 4. International Society for Soil Mechanics and Geotechnical Engineering, London, pp 2777–2872. https://doi.org/10.3233/978-1-60750-031-5-2777
- Mestre NC, Rocha TL, Canals M, Cardoso C, Danovaro R, Dell'Anno A, Gambi C, Regoli F, Sanchez-Vidal A, Bebianno MJ (2017) Environmental hazard assessment of a marine mine tailings deposit site and potential implications for deep-sea mining. Environ Pollut 228:169–178. https://doi.org/10.1016/j.envpol. 2017.05.027
- Miller KA, Thompson KF, Johnston P, Santillo D (2018) An overview of seabed mining including the current state of development, environmental impacts, and knowledge gaps. Front Mar Sci. https://doi.org/10.3389/fmars.2017.00418
- Moldoveanu N, Combee L, Egan M, Abriel W, Hampson G (2012) Over/under towed-streamer acquisition: a method to extend seismic bandwidth to both higher and lower frequencies. Seg Technical Program Expanded Abstracts 25(1):35–41. https://doi. org/10.1190/1.2369709

- Mora P, Tarantola A (1989) Seismic imaging. Springer, New York. https://doi.org/10.1007/0-387-30752-4_124
- Muñoz-Royo C, Peacock T, Alford MH, Smith JA, Le Boyer A, Kulkarni CS, Lermusiaux P, Mirabito C, Wang D, Adams E, Ouillon R, Breugem A, Decrop B, Lanckriet T, Supekar R, Rzeznik A, Gartman J (2021) Extent of impact of deep-sea nodule mining midwater plumes is influenced by sediment loading, turbulence and thresholds. Commun Earth Environ 2(1):1–16. https://doi.org/10.1038/s43247-021-00213-8
- Muñoz-Royo C, Ouillon R, El Mousadik S, Alford MH, Peacock T (2022) An in situ study of abyssal turbidity-current sediment plumes generated by a deep seabed polymetallic nodule mining preprototype collector vehicle. Sci Adv 8(38):1219. https://doi. org/10.1126/sciadv.abn1219
- Nagihara S, Brooks JM, Bernard BB, Summer N, Cole G, Lewis T (2002) Application of marine heat flow data important in oil, gas exploration. Oil Gas J 100(27):43–50. https://www.researchgate. net/publication/287539200_Application_of_marine_heat_flow_ data_important_in_oil_gas_exploration
- Nian TK, Guo XS, Fan N, Jiao HB, Li DY (2018) Impact forces of submarine landslides on suspended pipelines considering the lowtemperature environment. Appl Ocean Res 81:116–125. https:// doi.org/10.1016/j.apor.2018.09.016
- Nian TK, Guo XS, Zheng DF, Xiu ZX, Jiang ZB (2019) Susceptibility assessment of regional submarine landslides triggered by seismic actions. Appl Ocean Res 93:101964. https://doi.org/10.1016/j. apor.2019.101964
- Nian TK, Song XL, Zhao W, Jiao HB, Guo XS (2022) Submarine slope failure due to overpressure fluid associated with gas hydrate dissociation. Environ Geotech 9(2):108–123. https://doi.org/10. 1680/jenge.19.00070
- Pak SJ, Seo I, Lee KY, Hyeong K (2018) Rare earth elements and other critical metals in deep seabed mineral deposits: composition and implications for resource potential. Minerals 9(1):3. https://doi. org/10.3390/min9010003
- Pan GF, Hua ZG (1996) Distribution of manganese nodules and its relation to geoogic and geographic environment in the Clarion-Clipperton zone. Oceanologia et Limnologia Sinica (in Chinese). http://en.cnki.com.cn/Article_en/CJFDTOTAL-HYFZ1 99604010.htm
- Pan JP, Zhang DW, Yue LQ, Wang Y, Hu W (2006) Status of global offshore oil and gas exploration and development and its trends. China Min 11:1–4 (in Chinese). http://www.cqvip.com/qk/ 92839a/200611/23293990.html
- Paulus E (2021) Shedding light on deep-sea biodiversity—a highly vulnerable habitat in the face of anthropogenic change. Front Mar Sci 8:667048. https://doi.org/10.3389/fmars.2021.667048
- Peacock T, Ouillon R (2022) The fluid mechanics of deep-sea mining. Ann Rev Fluid Mech. https://doi.org/10.1146/annur ev-fluid-031822-010257
- Perez F, Sanchez LE (2009) Assessing the evolution of sustainability reporting in the mining sector. Environ Manag 43(6):949–961. https://doi.org/10.1007/s00267-008-9269-1
- Phaneuf D, Cote I, Dumas P (1999) Evaluation of the contamination of marine algae (seaweed) from the St. Lawrence River and likely to be consumed by humans. Environ Res Sect 80:175–182. https:// doi.org/10.1006/enrs.1998.3915
- Price DG (2008) Engineering geology: principles and practice. Springer. https://doi.org/10.1007/978-3-540-68626-2
- Rahman MS (1994) April. Instability and movements of oceanfloor sediments: a review. In: The fourth international offshore and polar engineering conference. OnePetro. https://onepetro.org/ IJOPE/article-abstract/27159/Instability-And-Movement-of-Oceanfloor-Sediments-A?redirectedFrom=fulltext#:~:text= Instability%20and%20movements%20of%20oceanfloor%20sed

iments%20are%20potentially,failure%20leading%20to%20oce anfloor%20instability%20and%20mass%20movements

- Sasaoka T, Karian T, Hamanaka A, Shimada H, Matsui K (2016) Application of highwall mining system in weak geological condition. Int J Coal Sci Technol 3(3):311–321. https://doi.org/10. 1007/s40789-016-0121-6
- Schriever W (1952) Reflection seismograph prospecing—how it started. Geophysics 17(4):936–942. https://doi.org/10.1190/1. 1437831
- Schultheiss PJ (1990) Pore pressures in marine sediments: an overview of measurement techniques and some geological and engineering applications. Mar Geophys Res 12(1):153–168. https://doi.org/ 10.1007/BF00310570
- Seed HB, De Alba P (1986) Use of SPT and CPT tests for evaluating the liquefaction resistance of sands. Use of in situ tests in geotechnical engineering. ASCE, pp 281–302. https://cedb.asce.org/ CEDBsearch/record.jsp?dockey=0048828
- Sha ZB, Zhang M, Zhang GX, Liang JQ, Su PB (2015) Using 4C OBS to reveal the distribution and velocity attributes of gas hydrates at the northern continental slope of South China Sea. Appl Geophys 12(4):555–563. https://doi.org/10.1007/s11770-015-0515-z
- Shan Z, Wu H, Ni W, Sun M, Wang K, Zhao L, Lou Y, Liu A, Xie W, Zheng X, Guo X (2022) Recent technological and methodological advances for the investigation of submarine landslides. J Mar Sci Eng 10(11):1728. https://doi.org/10.3390/jmse10111728
- Shano L, Raghuvanshi TK, Meten M (2020) Landslide susceptibility evaluation and hazard zonation techniques—a review. Geoenviron Disast 7:18. https://doi.org/10.1186/s40677-020-00152-0
- Sharma R (2001) Estimation of sediment properties during benthic impact experiments. Mar Georesour Geotechnol 19(4):269–289. https://doi.org/10.1080/106411901753335335
- Sharma R (2005) Deep-sea impact experiments and their future requirements. Mar Georesour Geotechnol 23(4):331–338. https:// doi.org/10.1080/10641190500446698
- Sharma R (2015) Environmental issues of deep-sea mining. Proc Earth Planet Sci 11:204–211. https://doi.org/10.1016/j.proeps.2015.06. 026
- Sharma R (2017) Deep-sea mining: resource potential, technical and environmental considerations. Springer. https://doi.org/10.1007/ 978-3-319-52557-0_1
- Sharma R (2022) Environmental issues of deep-sea mining. Proc Earth Planet Sci 11:204–211. https://doi.org/10.1016/j.marpol.2022. 105011
- Sharma R, Kodagali VN (1993) Influence of seabed topography on the distribution of manganese nodules and associated features in the Central Indian Basin: a study based on photographic observations. Mar Geol 110(1–2):153–162. https://doi.org/10.1016/ 0025-3227(93)90111-8
- Shen XH, Shen GD (2022) Analysis of geological characteristics and impact assessment of a foreign submarine gas pipeline engineering. Eng Technol Res 7(3):31–33 (in Chinese). https://www. cnki.com.cn/Article/CJFDTOTAL-YJCO202203010.htm
- Shi XF, Fu YZ, Li B, Huang M, Ren X, Liu J, Yu M, Li C (2021) Research on deep-sea minerals in China: progress and discovery (2011–2020). Bull Miner Petrol Geochem 42(2):305–318 (in Chinese). https://doi.org/10.19658/j.issn.1007-2802.2021. 40.022.
- Shinohara M, Yamada T, Ishihara T, Araya A, Kanazawa T, Fujimoto H, Uehira K, Tsukioka S, Omika S, Iizasa K (2015) Development of an underwater gravity measurement system using autonomous underwater vehicle for exploration of seafloor deposits. Oceans. https://doi.org/10.1109/OCEANS-Genova.2015.7271487
- Silcox WH, Bodine JA, Burns GE, Reeds CB, Wilson DL, Sauve ER (1987) Offshore operations. In: Petroleum engineering handbook, Chapter 18, 2nd edn. SPE, Richardson. https://www.osti.gov/biblio/5929149

- Simyrdanis K, Papadopoulos N, Cantoro G (2016) Shallow off-shore archaeological prospection with 3-D electrical resistivity tomography: the case of Olous (Modern Elounda), Greece. Remote Sens Basel. https://doi.org/10.3390/rs8110897
- Song LQ (1999) Geotechnical properties of oceanic polymetallic nodule sediments. Haiyangxuebao 6:47–54 (in Chinese). https:// www.cnki.com.cn/Article/CJFDTotal-SEAC199906005.htm
- Song SX (1993) Engineering geological conditions and mining methods of Jinchuan underground mine. China Mine Eng 3:2–10 (in Chinese). https://www.cnki.com.cn/Article/CJFDTOTAL-YSKS199303001.htm
- Sparenberg O (2019) A historical perspective on deep-sea mining for manganese nodules, 1965–2019. Extract Ind Soc 6(3):842–854. https://doi.org/10.1016/j.exis.2019.04.001
- Spearman J, Taylor J, Crossouard N, Cooper A, Turnbull M, Manning A, Lee M, Murton B (2020) Measurement and modelling of deep sea sediment plumes and implications for deep sea mining. Sci Rep 10(1):1–14. https://doi.org/10.1038/s41598-020-61837-y
- Stanton TE, Casagrande A, Housel WS, Tschebotareff GP (1939) Penetration devices—bearing. In: Proceedings of the eighteenth annual meeting of the highway research board held at Washington, D.C. November 28-December 2, 1938. Part II: Soil mechanics and soil stabilization, vol 18, pp 421–422. https://onlinepubs. trb.org/Onlinepubs/hrbproceedings/18/18Part2-044.pdf
- Stegmann S, Sultan N, Garziglia S, Pelleau P, Apprioual R, Kopf A, Zabel M (2012) A long-term monitoring array for landslide precursors: a case study at the ligurian slope (Western Mediterranean Sea). In: Offshore technology conference. OnePetro. https:// doi.org/10.4043/23271-MS
- Stewart DP, Randolph MF (1991) A new site investigation tool for the centrifuge. A A Balkema, Rotterdam. https://www.seman ticscholar.org/paper/A-new-site-investigation-tool-for-the-centr ifuge-Stewart-Randolph/9d13e147fd01e1dd0549794ba9428b 753eefb729
- Su Y (2020) Deep sea exploitation under equipment perspective. Wuhan University (in Chinese). https://doi.org/10.27379/d.cnki. gwhdu.2020.000706
- Sun HQ, Zhuang DD, Song CX, Zhu ZM, Wang F (2021) Geotechnical characteristics of seabed sedimelnts in the study area of polymetallic nodules in the east pacific ocean. J Ningbo Univ Technol 33(1):8–12 (in Chinese). https://www.nstl.gov.cn/paper_detail. html?id=db4559d667ebc75c63a1394302ba1176
- Sun ZW, Jia YG, Quan YZ, Liu T, Meng QS, Sun ZQ, Li K, Fan ZH, Chen T, Tang HR (2022) Development and application of longterm in situ monitoring system for complex deep-sea engineering geology. Earth Sci Front 29(5):216–228 (in Chinese). https://doi. org/10.13745/j.esf.sf.2021.9.15
- Sussman GJ, Stallman RM (1975) Heuristic techniques in computer aided circuit analysis. IEEE Trans Circuits Syst 22(11):857–865. https://doi.org/10.1109/tcs.1975.1083985
- Ted M (2014) Worldwide survey of deepwater drilling rigs [EB/OL]. (2014b-07-01). Offshore Magazine, Houston. https://www.offshore.mag.com
- Thiel H, Schriever G (1990) Deep-sea mining, environmental impact and the DISCOL project. Ambio 19(5):245–250. https://www. researchgate.net/publication/279903214_Deep-sea_mining_envir onmental_impact_and_the_DISCOL_project
- Torge W (1989) Gravimetry. Walter de Gruyter, Berlin. https://doi.org/ 10.1016/0031-9201%2891%2990144-7
- Vajk R (1949) Baron Roland Eotvos. Geophysics 14(1):6–9. https:// doi.org/10.1190/1.1437519
- Van Dover CL, Ardron JA, Escobar E, Gianni M, Gjerde KM, Jaeckel A, Jones DOB, Levin LA, Niner HJ, Pendleton L, Smith CR, Thiele T, Turner PJ, Watling L, Weaver PPE (2017) Biodiversity loss from deep-sea mining. Nat Geosci 10(7):464–465. https:// doi.org/10.1038/ngeo2983

- Vonnahme TR, Molari M, Janssen F, Wenzhöfer F, Haeckel M, Titschack J, Boetius A (2020) Effects of a deep-sea mining experiment on seafloor microbial communities and functions after 26 years. Sci Adv 6(18):5922. https://doi.org/10.1126/sciadv.aaz5922
- Wang JP, Wang CJ, Liu DH (2022) The establishment of environmental managementrules for exploitation of mineral resources in the international seabed areas and China's participation, 41(4): 610–618 (in Chinese). https://doi.org/10.13634/j.cnki.mes.2022. 04.018
- Weaver PPE, Aguzzi J, Boschen-Rose RE, Colaço A, de Stigter H, Gollner S, Haeckel M, Hauton C, Helmonsh R, Jones DOB, Lily H, Mestre NC, Mohn C, Thomsen L (2022) Assessing plume impacts caused by polymetallic nodule mining vehicles. Mar Policy 139:105011. https://www.sciencedirect.com/science/artic le/pii/S0308597X22000586
- Weaver PP, Billett DS, Dover CLV (2018) Environmental risks of deepsea mining. In: Handbook on marine environment protection. Springer, Cham, pp 215–245. https://doi.org/10.1007/978-3-319-60156-4_11
- Wei Z, He G, Deng X, Yao H, Liu Y, Yang Y, Ren J (2017) The progress in the study and survey of oceanic cobalt-rich crust resources. Geol China 44(3):460–472 (in Chinese). https://doi. org/10.12029/gc20170305
- Williams R, Erbe C, Duncan A, Nielsen K, Washburn T, Smith C (2022) Noise from deep-sea mining may span vast ocean areas. Science 377(6602):157–158. https://doi.org/10.1126/science. abo2804
- Xiao LJ, Fang M, Zhang WM (2000) Advance and present state of the research in oceanic metalliferous nodule mining. Metal Mine 8:11–14 (in Chinese). https://doi.org/10.3321/j.issn:1001-1250. 2000.08.004
- Xie H, Leung C, Wang J, Li X (2022) Advancing deep underground research through integration of engineering and science. Deep Underground Sci Eng 1(1):1–2. https://doi.org/10.1002/dug2. 12019
- Yang JM, Liu L, Lyu HN, Lin ZQ (2020) Deep-sea mining equipment in China: current status and prospect. Strateg Study CAE 22(6):1–9 (in Chinese). https://doi.org/10.15302/J-SSCAE-2020. 06.001
- Yu LC, King LS, Hoon ATC, Yean PCC (2015) A review study of oil and gas facilities for fixed and floating offshore platforms. Res J Appl Sci Eng Technol 10(6):672–679. https://doi.org/10.19026/ rjaset.10.2476
- Zhang FY, Zhang WY, Yin RG, Cheng YS, He GW (2001) A approach on the effect of manganese nodule resources associated with plate tectonics in the eastern part of the Pacific. Haiyang Xuebao 4:59–68 (in Chinese). https://doi.org/10.3321/j.issn:0253-4193. 2001.04.007
- Zhang H (2011) Research on magnetic exploration inversion technology and its application. Hebei University of Engineering, Handan (in Chinese). https://cdmd.cnki.com.cn/Article/CDMD-10076-1011162923.htm
- Zhang L (2015) OBS multi-component seismic data processing technology and its application. The University of Chinese Academy of Sciences, Guangzhou (in Chinese). http://www.irgrid.ac.cn/ handle/1471x/1136528
- Zhang H, Liu X, Chen A, Li W, Lu Y, Guo X (2021a) Design and application of an in situ test device for rheological characteristic measurements of liquefied submarine sediments. J Mar Sci Eng 9(6):639. https://doi.org/10.3390/jmse9060639
- Zhang M, Niu M, Shen S, Dai S, Xu Y (2021b) Review of natural gas hydrate dissociation effects on seabed stability. Nat Hazards 107(2):1035–1045. https://doi.org/10.1007/s11069-021-04629-5
- Zhang L, Zhou X, Zhou Y, Zhou J, Guo J, Zhai Z, Chen Y, Su X, Ying L, Wang L, Qiao Y (2022) Surface coal mining impacts on land

use change and ecological service value: a case study in Shengli coalfield, Inner Mongolia. Int J Coal Sci Technol 9(1):65. https://doi.org/10.1007/s40789-022-00518-9

- Zhu HC (2019) Research on submarine terrain deform-ation monitoring system based on MEMS sensing arrays and its data acquisition synchronization technology. Zhejiang University (in Chinese). https://doi.org/10.27461/d.cnki.gzjdx.2019.000219
- Zou HF, Liu SY, Cai GJ, Puppala AJ (2020) Probabilistic identification of contaminated soils using resistivity piezocone penetration

tests. Acta Geotech 15(3):761-779. https://doi.org/10.1007/ s11440-018-0708-8

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.